

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

He, Qing; Lu, Liping; Ongena, Steven

Working Paper Who gains from credit granted between firms? Evidence from inter-corporate loan announcements made in China

CFS Working Paper Series, No. 529

Provided in Cooperation with: Center for Financial Studies (CFS), Goethe University Frankfurt

Suggested Citation: He, Qing; Lu, Liping; Ongena, Steven (2016) : Who gains from credit granted between firms? Evidence from inter-corporate loan announcements made in China, CFS Working Paper Series, No. 529, Goethe University Frankfurt, Center for Financial Studies (CFS), Frankfurt a. M.,

https://nbn-resolving.de/urn:nbn:de:hebis:30:3-395767

This Version is available at: https://hdl.handle.net/10419/128500

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



CFS Working Paper Series

No. 529

Qing He, Liping Lu, Steven Ongena

Who Gains from Credit Granted between Firms? Evidence from Inter-corporate Loan Announcements *Made in China*

Center for Financial Studies | Goethe University | Theodor-W.-Adorno-Platz 3 | 60323 Frankfurt am Main | Germany Phone +49 69 798 30050 | Fax +49 69 798 30077 | info@ifk-cfs.de | www.ifk-cfs.de

Who Gains from Credit Granted between Firms? Evidence from Inter-corporate Loan Announcements *Made in China*

Qing He Renmin University of China

School of Finance Zhongguancun Avenue 59, 100872 Beijing, China Telephone: + 86 10 825 092 60 Email: qinghe@ruc.edu.cn

> Liping Lu VU University Amsterdam

Department of Finance De Boelelaan 1105, 1081HV Amsterdam, The Netherlands Telephone: +31 20 598 22 73, Fax: +31 20 598 60 20 E-mail: *l.lu@vu.nl*

> Steven Ongena University of Zurich, SFI and CEPR

Department of Banking and Finance Plattenstrasse 32 (PLD F-02), CH-8032 Zurich, Switzerland Telephone: + 41 44 634 39 54 Email: steven.ongena@bf.uzh.ch

This Draft: February, 2016

Electronic copy available at: http://ssrn.com/abstract=2734657

Who Gains from Credit Granted between Firms? Evidence from Inter-corporate Loan Announcements *Made in China*

Abstract

Who gains from inter-corporate credit? To answer this question we measure the impact of the announcements of inter-corporate loans in China on the stock prices of the firms involved. We find that the average abnormal return for the issuers of inter-corporate loans is significantly negative, whereas it is positive for the receivers. Issuing firms may be perceived by investors to have run out of worthwhile projects to finance, while receiving firms are being certified as creditworthy. Subsequent firm performance and investment confirms these valuations as overall accurate.

Key Words: Entrusted loan, inter-corporate loan, credit misallocation, certification JEL Classification: G30, G140, G210

1. Introduction

Credit between firms plays a crucial role in many economies around the world (Almeida and Wolfenzon (2006)). Firms with limited access to intermediated funds rely heavily on financial inter-linkages with other firms (Gopalan, Nanda and Seru (2007)). This is particularly important in emerging economies, where the legal systems are weak. The absence of adequate legal enforcement makes it burdensome for firms to raise external financing (La Porta, Lopez-de-Silanes, Shleifer and Vishny (1997)), which may lead to credit rationing of formal finance by financial institutions. Inter-corporate lending may be less subject to credit rationing and therefore may support the high growth in emerging economies like China (Allen, Qian and Qian (2005)).

Despite their ubiquity, research on inter-corporate credit continues to be hampered by a lack of direct firm-to-firm level data.¹ Thus, the inner workings of inter-corporate loans remain relatively unexplored. In this paper, we assemble a unique dataset to study the announcements of inter-corporate loans in the Chinese stock market during 2005-2012.² Indeed, as small- and medium-sized enterprises (SMEs) may face substantial obstacles in obtaining bank credit (Poncet, Steingress and Vandenbussche (2010)), the Chinese government has allowed firms to obtain credit from other non-financial firms under the coordination of financial institutions.

¹ Garcia-Appendini and Montoriol-Garriga (2014) study U.S. firms' liquidity positions and Boissay and Gropp (2014) study defaults on payments to suppliers in France. Jacobson and Von Schedvin (2015) and Ellingsen, Jacobson and von Schedvin (2016) study a dataset that contains 52 million trade credit contracts issued by 51 suppliers over 9 years to about 199,000 unique customers in Sweden. See also Petersen and Rajan (1997), Love, Preve and Sarria-Allende (2007), and Burkart, Ellingsen and Giannetti (2011), among others.

² Relying on similar data sources Allen, Qian, Tu and Yu (2016) examine the role played by inter-corporate loans in shadow banking, while Chen, Ren and Zha (2016) study monetary policy transmission and small bank risk-taking through the brokering of this lending.

These inter-corporate loans, also called "entrusted loans", are playing an increasingly important role in supplying credit to firms in China.³

The inter-corporate loans must be disclosed as a separate report of listed firms according to Chinese regulations.⁴ We can therefore observe the market reactions for inter-corporate loans, which is an important advantage over other studies. Analyzing stock market reactions to a corporate financing event can provide an immediate and comprehensive assessment for the valuation effect of such an event. We thereby break new ground in inter-corporate loan research providing novel evidence on the reactions of the stock prices to inter-corporate loan announcements for both the issuing and receiving firms. Furthermore, the announcements of inter-corporate loans enable us to glean specifics on the lending behavior involved, i.e., the relationship between lender and borrower, maturity, interest rate and collateral. To the best of our knowledge, there is no other published work has studied a dataset with such comprehensive information on inter-corporate loans.

Inter-corporate loans can be categorized as *inter-group* or *intra-group* loans, depending on whether the lender and borrower are both affiliated with the same business group. When the firm grants a loan to another firm outside the business group, it is called an inter-group loan. Due to weak legal enforcement of formal

³ According to the *China Securities Regulatory Commission* (CSRC), an entrusted loan is a type of loan in which the lender (i.e., the principal) extends credit to the borrower (i.e., the trustee) at specified amount, maturity, interest rate, and usage of the loan. Banks and other financial institutions only act as account managers who earn commissions but bear no default risk. Instead, the lending firm bears all the default risk. Entrusted loans amounted to 2.55 trillion RMB in 2013 (i.e., about \$400 billion) and accounted for 14.7 percent of the total amount of financing in the country. Data source: *People's Bank of China*. The increase in entrusted loans in 2013 was equivalent to nearly 30 percent of bank loans, which almost doubled the portion of 2012. The *Wall Street Journal* featured reports on entrusted loans in China on December 8th, 2011, and May 1st, 2014.

⁴ The CSRC requires all listed firms to announce major events which may influence their stock prices.

contracts, formal credit provided by financial institutions may be rationed. Inter-corporate lending, which in essence is a type of informal financing based on reputation and inter-firm relationships, may be less subject to rationing and may therefore support the high growth rates observed in an emerging economy like China (Allen, Qian and Qian (2005)). In this scenario, a firm has surplus financial resources, while the other one has an investment opportunity in need of external financing. The lending firm screens and monitors the borrowing firm as well as bearing the default risk. As suppliers could have information and enforcement advantages over financial institutions in providing credit to their own clients, inter-group loans could redistribute funds to more profitable firms. Thus, investors should react positively to the announcement of inter-group loans for the issuing firms. The issuance of inter-group loans, however, may also convey the information that the issuing firms are running out of worthy projects to finance, which may lead to negative market reactions for the issuing firms.

A firm may also lend to another firm within the same business group, which is then called an intra-group loan. These loans in effect funnel credit within the group's internal capital market. Stein (1997), for example, shows that an internal capital market can channel credit from less efficient projects to more efficient ones, and that the reallocation of credit within business groups is more common and important in countries with underdeveloped external capital markets. And indeed Buchuk, Larrain, Muñoz and Urzúa I. (2014) find that intra-group loans in Chile actually enhance firm investment and return and that, due to the country's strict regulation and disclosure requirements, such loans do not suffer from tunneling.⁵ Furthermore, Almeida, Kim and Kim (2015) show that internal capital markets of Korea business groups mitigate the negative effects on investment and performance during the Asian financial crisis. Thus, intra-group loans may be the outcome of an efficient credit reallocation decision within the business group, which may lead to a positive market reaction.

Gopalan, Nanda and Seru (2007) show that in India the internal capital market within business groups is a support mechanism for financially weaker firms to avoid default and thus the negative spillovers to the rest of the business group. Also, large recipients of intra-group loans have lower abnormal stock returns and operating performance in the subsequent one- and two-year period. Similarly, Khanna and Yafeh (2005) show that the internal capital market provides mutual insurance within business groups among affiliated firms in unstable periods in Japan, Korea and Thailand. As a result, the issuance of intra-group loans may signal financial distress for a group firm among the uninformed investors, which may lead to a negative market reaction.

To examine whether announcements of inter-group versus intra-group loans convey information to investors about these non-financial corporate lenders and borrowers, we hand-collect the announcements of 719 inter-corporate loans that take place between 2005 and 2012. There were 564 announcements made by the lenders and 155 by the borrowers. We find that the stock market reacts negatively to the

⁵ The internal capital market may also come with a potential dark side, which is the expropriation of minority shareholders due to a separation of ownership and control (Claessens, Djankov and Lang (2000); Jiang, Lee and Yue (2010); Johnson, La Porta, Lopez-de-Silanes and Shleifer (2000)) and the rent seeking behavior of divisional CEOs (Scharfstein and Stein (2000)). Fan, Jin and Zheng (2014) show the tradeoff between the negative and positive sides of the internal capital market in China, i.e., tunneling corporate resources versus alleviating credit constraints.

issuance of inter-corporate loans, and positively to their receipt. For example the two-day cumulative abnormal returns (CARs) on the stocks of the lending firms are on average equal to a (statistically significant) -42 basis points (bps), while the CARs on the stocks of the borrowing firms are equal to 91 bps.

Specifically, we find that an issuance of intra-group loans to subsidiaries is associated with negative CAR, possibly signaling (to uninformed investors) financial distress for a group firm. Our results are consistent with the argument that firms support other firms in the same business group (Almeida, Kim and Kim (2015)). Moreover, because the issuance of inter-group loans also generates negative CARs, issuing intra-group loans may similarly signal a lack of worthy projects for the lending firms to finance.

In contrast, the receipt of both intra-group and inter-group loans generates positive CARs, which indicates a type of certification of the borrowing firms by these non-financial corporate lenders. The announcement effects for the receipt of inter-corporate loans stands in pointed contrast to the findings in Bailey, Huang and Yang (2011) and Huang, Schwienbacher and Zhao (2012): They show that bank loan announcements in China result in negative abnormal returns for the borrowing firms. This may be due to banks' limited information and their well-known soft budget problem. Hence lenders of inter-corporate loans may have better private information and be less subject to social and political pressure to subsidize low-quality firms. Thus, receiving an inter-corporate loan in China may provide the type of certification similar to that associated with the receipt of a bank loan in the U.S. Besides, the receipt of intra-group loans from controlling shareholders is a type of corporate propping-up by controlling shareholders in emerging markets (Friedman, Johnson and Mitton (2003); Jian and Wong (2010); Peng, Wei and Yang (2011)), which leads to the positive market reactions.

Furthermore, we investigate the factors affecting the likelihood of issuing or receiving inter-corporate loans. Firms with higher market to book ratio are less likely to issue inter-group loans, while the issuance of intra-group loans does not depend on this ratio. In contrast, firms with higher market to book ratio are more likely to receive inter-group loans, while the receipt of intra-group loans does not depend on it. Besides, state-controlled firms are more likely to issue and receive intra-group loans while the case for inter-group loans, which suggests a potential credit misallocation by the state-owned enterprises.

We also find that the CARs on the issuance and receipt of inter-corporate loans are associated with certain loan, counter-party, and (loan-announcing) firm variables. These findings deepen our understanding of inter-corporate loan announcements. On the one hand, the issuance of inter-corporate loans generates lower CARs on loans with high interest rate spreads, on intra-group loans, and on loans to firms with higher market-to-book ratios and larger amounts of other accounts receivable (i.e., inter-corporate loans outstanding). On the other hand, the receipt of inter-corporate loans is associated with lower CARs for loans with a higher interest rate spread, from state-owned lenders, and to non-state controlled borrowers. The results support the hypothesis that the issuance of inter-corporate loans signals (to uninformed investors) a lack of worthwhile projects to finance for the issuing firms, while the issuance of intra-group loans conveys additional information of financial distress about the group firm. We further examine the impacts of inter-corporate loans on the long-run performance for issuers and receivers. Return on assets (ROA) declines after the issuance of inter-corporate loans in particular for firms with high market to book ratio and intra-group loans to subsidiaries, which confirms a lack of worthwhile projects to finance in the issuing firms and financial distress for the group firm. Specifically, ROA decreases when firms with high market to book ratio issue inter-corporate loans, while this is not the case for firms with low market to book ratio. Put differently, credit is misallocated when firms with high growth potential issue inter-corporate loans to others, while this is not so for firms with low growth potential. Besides, we also find weak evidence that ROA increases after the receipt of inter-corporate loans in particular for the intra-group loans from controlling shareholders, which confirms the receipt of inter-corporate loans as a certification for the borrowers and some corporate "propping-up".

We also examine corporate investment after inter-corporate loan announcements. The capital expenditure increases less for firms issuing than for firms receiving inter-corporate loans, in particular for firms with high market to book ratio and when it involves inter-group loans, which confirms the signaling of a lack of investment opportunities for the issuing firms. Besides, firms receiving inter-corporate loans invest more, in particular for intra-group loans from controlling shareholders, which confirms the certification by the corporate lenders and propping-up.

Our study contributes to the literature in the following ways. First, we provide a novel result on the market reaction to announcements of inter-corporate loans, which extends the literature on information production by non-financial firms as creditors alongside banks and non-bank financial institutions (Best and Zhang (1993); Billett,

Flannery and Garfinkel (1995)). We show that the receipt of inter-corporate loans from non-financial firms is associated with positive market reactions, which indicates a certification effect by these non-financial firms as creditors. Second, our study is also related with the literature on the internal capital market of business groups (Stein (1997); Gopalan, Nanda and Seru (2007)). We show that the issuance of inter-corporate loans by a listed firm to another firm within the same business group is associated with negative market reactions, which suggests that the uninformed investors perceive the use of the internal capital market as a signal of rescuing distressed group firms, and that the lending firm is running out of worthy projects to finance. Finally, our findings shed light on informal finance (Allen, Qian and Qian (2005)). We show that the issuance of inter-corporate loans to firms outside the business group (i.e., informal loans) is associated with weakly negative market reactions, which indicates that uninformed investors perceive these informal loans as unfavorable investments probably due to their lack of expertise in loan extension.

The rest of this paper is organized as follows. Section 2 introduces the Chinese financial system. Section 3 sets out our hypotheses. Section 4 discusses the data and methodology. Section 5 provides summary statistics, determinants of loan issuance and receiving, and event studies of the issuance and receipt of inter-corporate loans. Section 6 links CARs to a set of loan, counter-party and firm-specific characteristics. Section 7 shows the post-performance of inter-corporate loan announcements such as profitability and investment in the long-run. Section 8 concludes.

2. The financial system in China

The formal financial sector is dominated by banks in China (Allen, Qian and Qian (2005)) with a bank credit to GDP ratio (1.11) that is substantially higher than the average for the other countries in their sample (0.73). According to National Bureau of Statistics of China, banks provided 51.4 percent of the total financing for Chinese firms in 2013. The banking system in China consists of a central bank, four large (national) state-owned banks,⁶ three policy banks, 12 (regional) joint-stock banks, hundreds of local banks (i.e. city/rural commercial banks, rural cooperative banks, and village banks), and city/rural credit cooperatives, etc. There are also hundreds of branches and offices of foreign banks which can conduct a limited set of commercial banking activities in China (Allen, Qian, Zhang and Zhao (2013)). Generally speaking, the four large state-owned banks dominate the credit market in China. In particular state-owned banks are mandated to pursue social benefits and stability, and their credit allocation is often based on some "noisy" information about the borrowers and not on commercial judgment (Bailey, Huang and Yang (2011)). Moreover, small and private firms have limited credit histories and collateral, and will not receive a government bailout in case of default. Thus, banks favor lending to state-owned enterprises (SOEs) and large private firms, and discriminate against small and private firms in China (Firth, Lin and Wong (2008)). In sum, a majority of bank credit is channeled to the SOEs and large private firms, while small and private firms face substantial obstacles in obtaining external finance from the formal financial sector.

⁶ The four large stated-owned banks are Industrial and Commercial Bank of China, Bank of China, Construction Bank of China and Agriculture Bank of China.

The capital market, which mainly consists of a bond market and an equity market, is also relatively underdeveloped in China. The bond market remains under-developed until recent years, although corporate bonds were first issued already in 1986. The market value of newly issued bonds in China was only 1.74 percent of GDP at the end of 2012, and corporate bond issuance accounts for just 11.19 percent of total bond issuance in China. In contrast, the newly established Shanghai Stock Exchange and Shenzhen Stock Exchange have enjoyed rapid expansion since their founding in 1990. As measured by total market capitalization, both of these stock exchanges ranked in the world's top ten at the end of 2011. However, the combined stock market is still quite small compared to the banking system. The market capitalization-to-GDP ratio was 52 percent in 2011 in China, which is much lower than the U.S. Despite its rapid growth, the stock market does not play a proper role in the country, where insider trading and speculation are prevalent (He and Rui (2014)). For example, the turnover rates on the Shanghai and Shenzhen Stock Exchanges are 178.5 and 344.3 percent, respectively, which is higher than most industrial countries and may indicate widespread speculative trading (Allen, Qian, Zhang and Zhao (2013)). Moreover, the two stock exchanges were established so as to provide a new source of funding to SOEs and to reduce the financial burden of government bailouts. Up until 2005, about 80 percent of the (more than 1,100) listed enterprises were converted from SOEs in China, while a majority of listed firms are still owned or controlled by the government nowadays. The Chinese government's dual role as both regulators and shareholders reduces the effectiveness of the stock market in terms of resource allocation and risk diversification.

Informal financing, which has been growing rapidly in China (due to widespread state intervention and financial repression), channels credit from state-owned and large private firms to SMEs in support of a fast growth of the Chinese economy (Allen, Qian and Qian (2005); Bose (1998)). The formal financial sector is inefficient in allocating credit due to a severe information asymmetry and weak law enforcement. Informal financing channels based on reputation and relationship may be filling the gap thanks to their advantages in screening, monitoring, and enforcement (Stiglitz (1990); Arnott and Stiglitz (1991)).

One type of informal financing is the inter-corporate loan. Because direct lending activities among non-financial firms are prohibited in China, entrusted loans have moved in to facilitate inter-corporate lending.⁷ Under financial regulations in China, non-financial firms can extend credit to other firms via entrusted loans in a process that is coordinated by banks and other financial institutions (banks hereafter). Lenders and borrowers can negotiate loan terms subject to certain financial regulations regarding for example amount, interest rate, maturity, and purpose.⁸ Banks merely act as agents on behalf of the lenders and coordinate the loan procedures, i.e., the contract signing, loan withdrawals, and repayment, etc. However, banks do not bear any default risk for the entrusted loans, which are often treated as off-balance sheet items.

⁷ On March 8th, 1993, *People's Bank of China* (PBOC) promulgated administrative decrees on entrusted loans as regards financial trust companies. On April 5th, 2001, the PBOC released a regulation on entrusted loans, "*Issues on Commercial Banks' Provision for Launching Entrusted Loans*". For an overview of the evolution of financial regulation of entrusted loans in China, see Appendix 1 for a survey of the laws relating to entrusted loans.

⁸ Lending General Provisions by the People's Bank of China were formulated in accord with the Law of the Commercial Banks and other relevant laws on August 1st, 1996. Article 7 states that entrusted loans should comply with the Lending General Provisions.

Appendix 1 gives a timeline for the related laws and regulations (for what we will henceforth call inter-corporate loans).

The Chinese financial authorities imposed only a mild set of regulations on entrusted loans because of their beneficial effect on credit reallocation. The market for entrusted loans has witnessed a rapid expansion with the gradual liberalization of interest rates in China, and it has recently become a key source of financing.⁹ The interest rate ceiling for entrusted loans was abolished by the People's Bank of China (PBOC) in October 2004, which enabled lenders to negotiate freely with borrowers on interest rates. Appendix 2 shows that the market share of inter-corporate loans has been growing rapidly, accounting for 15 percent of total financing in 2013 (the second largest financing source after bank lending) and totaling 2.55 trillion RMB.

The rapid growth in entrusted lending has however generated substantial concern about the credit risks involved, especially in light of the gloomy prospects for the Chinese economy since the global financial crisis. SOEs and large private firms often have very limited investment opportunities (as they struggle with inefficient organizational structures, policy burdens, and overcapacity problems) but still have easy access to bank credit. This abundant credit can be channeled to private SMEs at interest rates above the bank lending rate.¹⁰ However, inter-corporate loans may also

⁹ The interest rates are under extensive regulation by the People's Bank of China (PBOC). PBOC set the benchmark interest rate along with a rate floor and rate ceiling. The interest rate is only allowed to vary within specified bounds. For example, PBOC set the benchmark lending rates, and the interest rate of commercial loans, including entrusted loans, must be between the floor and ceiling around the benchmark lending rate. China began its interest rate liberalization in 1996 by abolishing the ceiling on interbank lending rates. From 1998 to 2004, the ceiling for the lending rates gradually raised, and was abolished in October 2004 (except for credit cooperatives), while the floor remained unchanged at 90% of the benchmark lending rate. Recently, China took a further step toward a market-oriented rate by removing the lending rate floor on July 19th, 2013.

¹⁰ Beijing, June 25th, 2013 (*Reuters*) - A deputy general manager in a state-owned steel firm says that the firm doesn't use the bank credit to expand production, as the average loss is 100 - 200 RMB per ton of

be extended to poor borrowers if lenders do not conduct proper screening and monitoring due to a lack of expertise in lending; e.g., 28 percent of outstanding inter-corporate lending by the *Sunny Loan Top Co., Ltd* as at end-2013 could not be recovered at maturity.¹¹ The risks of inter-corporate lending can increase the systemic risks of the financial system as such credit often ends up in the real estate market and local municipal government investment platforms, which have become a major concern as regards the financial stability in China.

Although inter-corporate loans have come under substantial scrutiny in China itself, the gains and losses from such loans have not been assessed yet in the academic literature. We will provide such an evaluation by examining the market reactions to the announcements of issuance and receipt of inter-corporate loans in the following sections, and also their impact on the profitability and corporate investment ex post in the long run.

3. The hypotheses

Institutional lenders, such as banks, can enhance firm valuation by alleviating the information asymmetry of borrowers (Fama (1985); Boot (2000); Ongena and Smith (2000)). Approval of a bank loan is often perceived by uninformed investors as a good signal, especially for borrowers who suffer from severe information asymmetries. The positive excess returns on borrowers' stocks following bank loan announcements are

steel sold. Entrusted loans are an attractive business option for his company. The firm borrows from banks at the benchmark lending rate (about six percent), and issues entrusted loans to borrowers at twice that rate.

¹¹ The firm is listed in *Shanghai Stock Exchange* (i.e. stock ID: 600830). Among 1.12 billion RMB entrusted loans outstanding at the end of 2013, 306 million is classified as doubtful and 5 million is classified as losses. Data from the announcement by *Sunny Loan Top Co., Ltd*: www.cninfo.com.cn/finalpage/2014-06-07/64111840.PDF.

widely documented in the literature. For example, James (1987) finds positive CARs of almost 200 basis points in a two-day period surrounding bank loan announcements.¹² Billett, Flannery and Garfinkel (1995) show significant positive CARs on loans from non-bank financial institutions which are indistinguishable from bank loans. However, Lummer and McConnell (1989) show that the market reacts positively to loan renewals but not to new bank loan agreements.

The CARs on bank loan announcements are higher for borrowers who suffer more from information asymmetries (Best and Zhang (1993)), for example these could be the smaller borrowers (Maskara and Mullineaux (2011)). Certain types of lenders can also better alleviate the information asymmetry of bank loan agreements, as in the case of internationally syndicated loans in emerging economies (Harvey, Lins and Roper (2004)); for example also foreign or local banks, except for domestic banks that are located far from their borrowers, may play that role (Ongena and Roscovan (2013)); or lenders with higher credit ratings (Billett, Flannery and Garfinkel (1995)).

However, the CARs on bank loan announcements change over time. Fields, Fraser, Berry and Byers (2006) for example show that CARs on bank loan announcements were positive in the 1970s and 1980s, whereas they disappeared afterwards except for smaller and poorly performing firms and periods of high credit risk spreads, a result also present in samples studied by Andre, Mathieu and Zhang (2001) and Ongena, Roscovan, Song and Werker (2014) for example. And Li and Ongena (2015) find that the CARs on bank loan announcements were positive during

¹² Following work by Mikkelson and Partch (1986). James and Smith (2000) and Degryse, Kim and Ongena (2009) provide a critical review of the methodology and the extant empirical evidence.

the global financial crisis although they were close to zero before that. They surmise that in a booming credit market the certification of corporate borrowers by banks started to play a lesser role, while during the crisis the banks' role was revitalized. Wang and Xia (2014) show that banks exert less effort in ex-ante screening and ex-post monitoring when they can securitize loans, which may also explain the changes in CARs on bank loans over the time.

Despite considerable anecdotal evidence, little direct evidence has been provided so far on the market reactions of lenders' stocks to loan announcements. A few studies investigate the loan announcement effect of lending financial institutions. For example, Megginson, Poulsen and Sinkey (1995) show that the announcements of syndicated loans to Latin American borrowers in the 1970s are associated with negative CARs for the lending banks, while syndicated loans to U.S. borrowers in the 1980s are associated with positive CARs. However, little is known about the announcement effect of loans made by non-financial firms. Yook (2003) show that the acquirers' stock prices suffer from negative market reactions to M&A announcements, which may be seen to indicate that the acquirers have run out of other worthwhile projects to finance. Similarly, the issuance of inter-corporate loans may suggest that the issuing firms have run out of worthy investment projects, which can lead to negative market reactions. Thus, our first hypothesis is:

Hypothesis 1: The announcement of the issuance of an inter-corporate loan will lead to significantly negative excess returns on the stock of the issuing firm.

In contrast to bank loans, the granters of inter-corporate loans are non-financial firms, and loan default risks should incentivize them to acquire proprietary information about the borrowers, e.g., through long-term business relationships such as business group affiliation, supplier-customer relationship, or personal relationship of the CEOs, etc. Thus, obtaining an inter-corporate loan may certify the borrower and convey positive information to uninformed investors. Thus, our second hypothesis is:

Hypothesis 2: The announcement of the receipt of an inter-corporate loan will lead to significantly positive excess returns on the stock of the receiving firm.

Business groups are prevalent in emerging economies (Claessens, Fan and Lang (2006)), where weak creditor protection may make it too costly to raise external finance (Bae and Vidhan (2009)). Firms with good investment opportunities can obtain credit through the internal capital market if and when the headquarters of the business group allocates credit efficiently among group firms (Stein (1997)). But the headquarters of the group may also channel credit to other group firms in financial distress in order to avoid negative spillovers to the rest of the business group (Gopalan, Nanda and Seru (2007); Almeida, Kim and Kim (2015)).

Ownership is often concentrated in business groups, and voting rights exceed cash flow rights through pyramid structures and cross-holdings by controlling shareholders in emerging economies (Claessens, Djankov and Lang (2000)). Thus, a majority of decision rights are often in the hands of controlling shareholders, which may enable them to tunnel corporate resources for private benefits (Johnson, La Porta, Lopez-de-Silanes and Shleifer (2000)). For example, controlling shareholders used to tunnel resources away from listed firms through intra-group loans in China (Jiang, Lee and Yue (2010)). However, intra-group loans to controlling shareholders have been prohibited by the Chinese government since 2006, which alleviates the tunneling concerns with respect to inter-corporate loans in the country. As a result, intra-group loans among Chinese listed firms are more likely to reveal a lack of worthwhile projects in the issuing firms and financial distress of group firms rather than tunneling.

The receipt of intra-group loans, however, resembles bank loan agreements though the lenders are affiliated within the same business group. Firms may have proprietary information due to affiliation within the same business group. Furthermore, intra-group loans from controlling shareholders are a type of corporate propping-up (Friedman, Johnson and Mitton (2003); Jian and Wong (2010); Peng, Wei and Yang (2011)), which may generate positive market reactions. Thus, our third and fourth hypotheses are:

Hypothesis 3: The announcement of the issuance of an intra-group loan will lead to significantly negative excess returns on the stock of the issuing firm.

Hypothesis 4: The announcement of the receipt of an intra-group loan will lead to significantly positive excess returns on the stock of the receiving firm.

Inter-group loans are extended by one non-financial firm to another not in the same business group. But inter-group loans often occur between firms with certain business relationships, e.g., customers, suppliers, or firms in the same industry. Lending firms may have an informational advantage over banks in screening and monitoring the borrowers when extending inter-corporate loans. In addition, lenders can benefit from inter-group loans via a higher rate of return than in alternative investments (e.g., bank deposits). What is even more important is that inter-group loans provide an alternative financing channel for credit constrained borrowers such as private SMEs.

Lenders of inter-group loans, especially SOEs and large private firms, can raise external finance more easily than private SMEs. The issuance of inter-group loans may signal to uninformed investors that the issuing firms have run out of worthwhile projects to finance, even though the inter-corporate loans indeed improve the credit allocation in the issuing firms as in Bose (1998) and Hoff and Stiglitz (1997). We surmise that the signaling effect dominates the credit reallocation effect, which results in a negative market reaction to the issuance of inter-group loans.

The receipt of inter-group loans, however, is also quite similar to the receipt of bank loans in terms of certification for the borrowers. Although the lenders are non-financial firms which may lack sufficient expertise in lending, they may have proprietary information obtained through long-term business relationship (e.g., suppliers or customers). The receipt of inter-corporate loans may reveal proprietary information to uninformed investors, which can generate positive market reactions. Thus, our fifth and sixth hypotheses are: *Hypothesis 5: The announcement of the issuance of an inter-group loan will lead to significantly negative excess returns on the stock of the issuing firm.*

Hypothesis 6: *The announcement of the receipt of an inter-group loan will lead* to significantly positive excess returns on the stock of the receiving firm.

Bailey, Huang and Yang (2011) show that the announcements of bank loans are associated with negative abnormal returns on the borrower's stock, and obtaining bank loans predicts poorer subsequent performance of borrowers in China. Banks do not screen or monitor borrowers in a proper way, which results in inefficient lending and thus credit misallocation in the country. In contrast, inter-corporate loans based on frequent interactions between borrowers and lenders are more likely to be driven by business considerations, which may reveal proprietary information to the uninformed investors and also provide alternative financing for high growing but credit constrained firms (e.g., intra-group loans from controlling shareholders as corporate propping-up). Hence, corporate performance and investment should be higher after the receipt of inter-corporate loans, which echoes with the positive market reaction to the loan receipts. In contrast, it may reveal a lack of worthwhile projects to finance in the issuers when issuing inter-corporate loans, and may also reveal potential financial distresses in group firms when issuing intra-group loans to subsidiaries. Thus, we propose that firms issuing inter-corporate loans will have lower corporate performance and investment, and vice versa for firms receiving inter-corporate loans. Thus, our seventh and eighth hypotheses are:

Hypothesis 7: The issuance of an inter-corporate loan will lead to a deterioration of corporate performance and investment in the issuing firm.

Hypothesis 8: The receipt of an inter-corporate loan will lead to an increase in corporate performance and investment of the receiving firm.

4. Data and variables

Our sample consists of non-financial firms traded on the Chinese stock market (both *Shanghai* and *Shenzhen Stock Exchange*). We firstly identify a sample of 1,024 announcements of inter-corporate loans during 2005-2012 from *Resset* (www.resset.cn), which is a widely used database for the Chinese stock market (Calomiris, Fisman and Wang (2010)). The CSRC requires all listed firms to announce major events which may influence their stock prices.¹³ We then crosscheck the announcements with the official documents of corporate announcements

¹³ According to Article 67 of Chapter 3 of the *Securities Law of China* (effective as of Oct 27, 2005), the term "major event" means: (1) A major change in the company's business guidelines or scope of business; (2) A decision made by the company concerning a major investment or major asset purchase; (3) Conclusion by the company of an important contract which may have an important effect on the company's assets, liabilities, rights, interests or business results; (4) Incurrence by the company of a major debt or default on an overdue major debt; (5) Incurrence by the company of a major deficit or incurrence of a major loss; (6) A major change in the external conditions of the company's production or business; (7) A change in the board of directors, no less than one-third of directors, supervisors or managers of the company's shares; (9) A decision made by the company to reduce its capital, to merge, to divide, to dissolve, or to apply for bankruptcy; (10) Major litigation involving the company, or lawful cancellation by a court of a resolution adopted by the shareholders' general meeting or the board of directors; (11) Criminal cases involving the company, and the arrest of board of directors, supervisors or senior management staff; (12) Other events specified by the *China Securities Regulatory Commission*.

published on the websites designated by the CSRC,¹⁴ and the websites of the *Shanghai* and *Shenzhen* Stock Exchanges. We identify another 249 announcements of inter-corporate loans. So we reach a sample of 1,273 announcements of inter-corporate loans. Appendix 3 shows an example of an inter-corporate loan announcement record (translated by the authors).

We exclude all observations that coincide with other confounding corporate events (i.e., release of annual reports, announcement of seasonal offerings, dividend, law suits, etc.) within the [-2, 2] trading day window around the announcements date of an inter-corporate loan. We obtain a sample of 719 unaffected announcements of issuance and receipt of inter-corporate loans. Appendix 4 tabulates the total volume of inter-corporate loans by all listed firms on *Shanghai Stock Exchange*, which shows that our sample covers a quarter of the total volume of inter-corporate loans.

We record the announcement date, loan type (i.e., issuance/receipt), existing relationship between lender and borrower (i.e., inter-group and intra-group, where the latter is further broken down into controlling shareholders, subsidiaries, and firms with other relationships (i.e., firms affiliated with the same business group but without equity ownership of each other), and ownership of the counter-party. In addition, we also record whether an inter-corporate loan is a new loan (issuance/receipt) or a loan revision. A new loan indicates that the borrower and lender do not have a prior inter-corporate loan between them, while a loan revision means there is an existing loan. The announcement files for inter-corporate loans enable us to identify loan terms

¹⁴ The official designated websites for corporate disclosures are www.cninfo.com.cn and www.cnstock.com.

such as the loan amount, interest rate, maturity, and collateral, and also the name of the financial institution involved, among other characteristics.

The inter-corporate loan announcements are matched with stock prices and a set of firm characteristics at the fiscal year-end before the announcement year. We collect financial information for non-listed firms from the announcement files of inter-corporate loans, and also from the survey of industrial firms by the *National Bureau of Statistics of China*.¹⁵ As a result, we can compile a set of firm characteristics which may be associated with the CARs on announcements of inter-corporate loans.

We include loan variables in the regression (loan size, interest rate, maturity, guarantee, and loan revision), as well as counter-party variables (intra-group and inter-group counter-party, counter-party industry, counter-party size, and state-owned counter-party). In addition, we include a set of firm variables: Size, age, market-to-book ratio, free cash flow, leverage, state-control, other accounts receivable, and other accounts payable. Finally, we include industry and year fixed effects in the regression. Variable definitions are listed in Table 1.

[Table 1 here]

¹⁵ We use the dataset for industrial firms in China, which include all state-owned enterprises (SOEs) and all non-state owned firms with annual sales revenues above five million RMB, from 1998-2009.

5. Results

5.1 Summary statistics

We focus on 2005-2012, as our sample covers all announcements of inter-corporate loans in that period. Panel A of Table 2 shows the distribution of 719 "clean" announcements by type and year. A total of 564 announcements were made on the issuance of inter-corporate loans, and 155 announcements on their receipts. The number of announcements increases over the years, with a slight decrease in 2012, and there are more announcements of issuances than of receipts.

[Table 2 here]

Panel B of Table 2 shows the distribution of inter-corporate loan announcements by industry. A majority of the inter-corporate loans are in the manufacturing industry. The utilities industry ranks second in the number of issuance announcements of inter-corporate loans, whereas the real estate industry ranks second for receipt announcements.

Panel A of Table 3 shows the characteristics of inter-corporate loans for issuance versus receipts as well as for intra-group versus inter-group loans. A majority of the announced inter-corporate loans are intra-group loans, on both issuance and receipts. Panel B of Table 3 shows a further decomposition of the intra-group loans into controlling shareholders, subsidiaries, and firms with other relationships. All issuances of intra-group loans go to the subsidiaries of listed firms except for four loans to the controlling shareholders, while a majority of the receipts of intra-group loans comes from the controlling shareholders. Table 3 also shows that issuance size (in million RMB) is smaller than the receipt size for inter-corporate loans, the

respective average loan sizes for issuance and receipt of inter-corporate loans being 185 and 300 (217 and 298 for intra-group loans).

[Table 3 here]

Table 3 also shows that the maturity is 18 months on average for the issuance of intra-group loans and 12 months for the issuance of inter-group loans. The maturity for the receipt of inter-corporate loans is slightly longer than that for inter-group loans. In addition, the average interest rate spread is 14 versus 115 percent over the basis lending rate for the issuance of intra-group versus inter-group loans; and 4 versus 17 percent for the receipt of intra-group versus inter-group loans. Furthermore, the issuance of intra-group loans has a lower proportion of guarantees than inter-group loans (i.e., 13 versus 66 percent), which also applies to the receipt of intra-group versus inter-group loans (i.e., 9 versus 40 percent).

5.2 Determinants of issuing and receiving inter-corporate loans

In order to identify factors that may affect the likelihood of issuing or receiving inter-corporate loans, we use size-based matching method in Beasley (1996) and Bailey, Huang and Yang (2011). It allows us to identify a group of control firms with similar size and same industry. The dependent variable equals 1 for each firm-year in our sample, and 0 for a matching sample constructed for each firm-year from all firms that do not have any record on issuing or receiving inter-corporate loans. Specifically, for each firm-year in our loan announcement sample, we identify all other firms from the same industry and choose the one with the closest value of total assets, as long as it is within the ten percent band of the sample firms' total assets. We then pool these

matched firms with our sample firms, and run a regression of the likelihood of issuing or receiving inter-corporate loans on a set of firm characteristics.

Models (1)-(3) of Table 4 show that more mature firms are more likely to issue inter-group loans, while less likely for intra-group loans. In addition, firms with higher market to book ratio is less likely to issue inter-group loans, while it doesn't matter for the issuance of intra-group loans. Furthermore, firms with higher leverage are less likely to issue inter-group loans, while it does not matter for the issuance of intra-group loans. Besides, state-controlled firms are more likely to issue intra-group loans, while this is not the case for inter-group loans. What is more, firms with a CEO who is also a chairman of the board of directors are less likely to issue intra-group loans, which reflects the relative bargaining power in terms of credit re-allocation between the headquarter CEO and divisional CEO of a business group.

[Table 4 here]

Models (4)-(6) of Table 4 show that more mature firms are less likely to receive inter-group loans. Also, firms with higher market to book ratio are more likely to receive inter-group loans, while market-to-book does not matter for the receipt of intra-group loans. In addition, firms with higher leverage are less likely to issue inter-group loans but more likely to receive inter-group loans. Furthermore, state-controlled firms are more likely to issue and receive intra-group loans, while not so for inter-group loans. Firms with higher profitability, however, are less likely to receive inter-group loans, which may reflect abundant cash flow in the firm thus a lower external credit demand.

5.3 Market reactions to the issuance of inter-corporate loans

A standard market model (as in Thompson (1985)) is used to estimate the benchmark returns and then to calculate the abnormal returns. In order to measure market returns, we use the equally-weighted market return for the Chinese stock market (A-shares) from the China Stock Market and Accounting Research (CSMAR) database. We define the announcement date as the event date (i.e., "day 0"). For each "clean" announcement of entrusted loans, we run a daily market model for the firms over the estimation window of [-250, -21], and calculate abnormal returns in the event windows accordingly.

Since December 16, 1996, the Chinese government has imposed restrictions on the ceiling and floor of the daily stock price. Based on previous trading day's closing price, the ceiling and floor for the stock prices are set at ten percent for all stocks and five percent for stocks that are labeled as *special treatment status* ("ST").¹⁶ Thus, the stock price may continue to react after the announcement day, which makes CAR[-1, +1] an informative measure to capture a full market reaction besides the standard CAR[-1,0]. We also report results for various event windows (e.g., CAR[-2,+2]) to check the robustness.

The top panel of Table 5 shows the abnormal returns on the issuance of inter-corporate loans. The two-day cumulative abnormal return, i.e., CAR[-1,0], is

¹⁶ According to CSRC, a company can be downgraded to ST status if: (1) The firm records a net loss in two consecutive fiscal years; (2) The company is found to have committed financial fraud and, after taking remedial action, records a net loss in two consecutive fiscal years; (3) The company is found to have committed financial fraud, the company has failed to take remedial action within a specified period after being urged by the CSRC to do so, and the company has been temporarily delisted for two months; (4) The company has failed to issue its annual report or semi-annual report on the designated date and has been temporarily delisted for two months. Any company that fails to take steps to improve its situation after being designated ST will ultimately be delisted from the stock exchange.

-0.42 percent and statistically significant at the one percent level in a Student's t-test and significant at the one percent level in a Wilcoxon signed-rank test (i.e., the proportion of positive CARs is less than 50 percent). The results for AR[0] and CAR[-1,1] are qualitatively similar. The upper panel of Appendix 5 shows the average abnormal returns for the issuance of inter-corporate loans in the [-20, 20] window, which exhibits a substantial drop in the average abnormal return on the event day.

[Table 5 here]

As multiple inter-corporate loan issuances may be announced on a single day, we split the sample into single-event versus multiple-events subsamples in order to get a cleaner effect. Thus, the middle panel of Table 5 shows that the CAR[-1,0] is -0.48 percent and statistically significant at the one percent level for the single-event subsample in both the t-test and Wilcoxon signed-rank test, and -0.17 percent and insignificant for the multiple-events subsample. We will focus on the single-event subsample for the issuance of inter-corporate loans hereafter.

Table 6 shows the issuance of inter-corporate loans by intra-group versus inter-group loans. The CAR[-1,0] for the issuance of intra-group loans is -0.62 percent and significant at the one percent level in both a t-test and a Wilcoxon singed-rank test, while it is -0.25 percent for inter-group loans and significant at one percent level in a Wilcoxon signed-rank test although statistically insignificant in a t-test. Furthermore, the results for AR[0] and CAR[-1,-1] are qualitatively similar. In sum, the issuance of inter-corporate loans generates a negative market reaction, sending unfavorable signals to uninformed investors for both intra-group and inter-group loans. The issuance of inter-corporate loans may reveal a credit misallocation in a listed firm, e.g.,

a listed firm may run out of worthwhile projects to finance even when the inter-corporate loans may enhance the credit allocation.

[Table 6 here]

We also show the CARs separated into issuance of intra-group loans to controlling shareholders, subsidiaries, and borrowers with other relationships. Table 7 shows that the issuance of intra-group loans to subsidiaries constitutes a majority of the sample, while the set of loans issued to controlling shareholders contains only four observations due to government sanctions since 2006. Consistent with the tunneling of intra-group loans to controlling shareholders in Jiang, Lee and Yue (2010), the CAR[-1,0] equals -1.97 percent, though it is not statistically significant due to the limited number of observations. In addition, CAR[-1,0] is -0.65 percent for the issuance of intra-group loans to subsidiaries, which is significant at the one percent level in both a t-test and a Wilcoxon signed-test. We do not find any significant CARs for the issuance of intra-group loans to borrowers with other relationships.

[Table 7]

Besides indicating a lack of worthwhile projects in the issuing firm, the issuance of intra-group loans to subsidiaries may also reveal financial distress in this subsidiary, which may spill over to the rest of the business group. Our results provide evidence consistent with the role of ICMs in supporting financially weaker firms in a business group (Gopalan, Nanda and Seru (2007); Almeida, Kim and Kim (2015)).

In sum, the negative CARs for the issuance of inter-corporate loans may reveal a credit misallocation to the uninformed investors, or rather a lack of worthwhile projects to finance for the issuing firms, even when such loans would improve credit

allocation. Furthermore, the negative CARs for the issuance of intra-group loans to subsidiaries may reveal financial distress in the subsidiary which may spill over to the rest of the business group. The negative CARs for the issuance of inter-group loans, however, may also show inefficient informal lending, i.e., non-financial corporate lenders may lack sufficient lending expertise as banks.

5.4 Market reactions to the receipt of inter-corporate loans

Table 8 shows the abnormal returns for the receipt of inter-corporate loans. The CAR[-1,0] is 0.91 percent and statistically significant at the one percent level in a t-test, and also at the ten percent level in a Wilcoxon signed-rank test. In addition, CAR[-1,0] is 1.09 percent and statistically significant at the one percent level in a t-test and at the five percent level for the Wilcoxon signed-rank test for the receipt of inter-corporate loans in the single-event subsample, and -0.59 percent in the multiple-events subsample (but not significant). Similarly, we will focus on the single-event subsample for the receipt of inter-corporate loans hereafter. The results are qualitatively similar for other event windows, such as AR[0], CAR[-1,1], and CAR[-2,2]. The lower panel of Appendix 5 shows a substantial jump in the average abnormal return on the event day of the receipt of inter-corporate loans. In sum, the receipt of inter-corporate loans in the U.S.

[Table 8 here]

We further tabulate the receipts of inter-corporate loans by intra-group versus inter-group loans. The top panel of Table 9 shows that the CAR[-1,0] for the receipt of intra-group loans is 1.06 percent and statistically significant at the one percent level in

a t-test and also at the five percent level in a Wilcoxon signed-rank test. The CAR[-1,0] is 1.33 percent for inter-group loans (but not statistically significant). The results are qualitatively similar for other event windows, e.g., AR[0], CAR[-1,1], and CAR[-2,2]. We find that the intra-group loans also have a certification effect for the borrowing firms, which may be due to their proprietary information due to affiliation with the same business group.

[Table 9 here]

We also tabulate the receipts of intra-group loans by those from controlling shareholders, subsidiaries, and lenders with other relationships. Table 10 shows that intra-group loans from controlling shareholders constitute a majority of the sample. The CAR[-1,0] is 1.42 percent for the receipt of intra-group loans from controlling shareholders and is statistically significant at the one percent level in both a t-test and a Wilcoxon signed-rank test. Furthermore, the receipt of intra-group loans from lenders with other relationship has a CAR[-1,0] of 0.37 percent (but not statistically significant). However, we do not find any clear evidence for the receipt of intra-group loans from subsidiaries, likely due to the limited number of observations.

[Table 10 here]

In sum, the receipt of inter-corporate loans may provide certification for the borrowing firms. In particular, the receipt of loans from controlling shareholders can provide information to uninformed investors, which leads to positive CARs. The intra-group loan from controlling shareholders is also consistent with the corporate propping-up by controlling shareholders in emerging markets (Friedman, Johnson and Mitton (2003); Jian and Wong (2010); Peng, Wei and Yang (2011)).

6. CARs and loan, counter-party and firm characteristics

We tabulate CARs in terms of sample median and by loan type, counter-party and firm characteristics in Table 11. The issuance of intra-group loans to subsidiaries generates higher CARs for loans with a guarantee, and lower CARs for loan revisions, and also lower CARs for those where the issuers have higher other accounts receivable (i.e., more inter-corporate loans outstanding). For the receipt of intra-group loans from controlling shareholders, the CARs are higher when the receiving firms have higher market to book ratio (though insignificant), higher ROAs and lower free cash flow.

[Table 11 here]

We also regress the CARs on loan, counter-party, and firm level variables. Firstly, we include loan variables, i.e., loan size, spread of the interest rate over basis lending rate, maturity, guarantee status (whether a loan is collateralized or guaranteed by third-parties), loan revision (whether a loan announcement relates to a revision of loan terms such as extension of maturities). We also include a set of counter-party variables, i.e., intra-group versus inter-group loans, whether the counter-party and the listed firm are in the same industry, counter-party size, and whether the counter-party is state-owned, etc. Finally, we include a set of firm level variables, i.e., firm size, age, market-to-book ratio, free cash flow, leverage, state control, other accounts receivable, and other accounts payable, etc. The balance of the issuance of inter-corporate loans is typically recorded in other accounts receivable, and while their receipt is recorded in other accounts receivable, and Yue (2010) show that other accounts receivable captures the tunneling activities of controlling shareholders. Because this channel has been closed since 2006, only four inter-corporate loans were issued to

controlling shareholders. We include other accounts receivable and payable to measure the balance of inter-corporate loans outstanding at the fiscal year-end before these announcements.

Appendix 6 shows summary statistics for the variables in the regression. Firms announcing the issuances of inter-corporate loans are more likely to be larger, more mature, state-controlled, and to have lower market to book ratio, lower leverage ratio, and lower other accounts receivable (i.e., lower inter-corporate loan issuances in the past) than those without any inter-corporate loan announcement. However, firms issuing inter-corporate loans are more likely to have lower free cash flow, which shows that the issuing firms may not simply relocate abundant cash for higher investment returns, e.g., rescuing group firms even though the list firm does not have abundant cash. In contrast, firms announcing the receipt of inter-corporate loans are more likely to be larger, more mature, state-controlled, and have higher leverage ratio, and higher other accounts payable (i.e., higher inter-corporate loans receipts in the past).

Table 12 shows the regressions of CAR[-1,0] for the issuance and receipt of inter-corporate loans on the loan, counter-party and firm characteristics. Models (1) to (3) give the estimates for the issuance, and Models (4) to (6) for the receipt of inter-corporate loans. Model (1) of Table 12 shows that the CARs on the issuance of inter-corporate loans are negatively associated with the interest rate spread over the basis lending rate. Model (2) shows that CARs on the issuance of inter-corporate loans are lower for intra-group loans, which may be due to a signaling effect as to the financial distress of a group firm. Specifically, intra-group loans to subsidiaries at a

higher interest rate may reveal a high default risk in these subsidiaries, which could spill over to the rest of the business group.

[Table 12 here]

Model (3) of Table 12 shows that the more mature firms in the stock market generate higher CARs on the issuance of inter-corporate loans, as the uninformed investors may already know their low growth potential, and also be less concerned about default risks for the borrowing subsidiaries. Furthermore, firms with higher market-to-book ratios have lower CARs due to more severe credit misallocation in the fast-growing firms. In other words, an issuance of an inter-corporate loan by a firm with higher growth prospects may provide more valuable proprietary information to the uninformed investors on a lack of worthwhile projects to finance in the issuing firm. Finally, the CARs are lower for issuers with higher other accounts receivable, i.e., the issuing firms already have substantial issuances of inter-corporate loans outstanding, which may further confirm credit misallocation in the issuing firm, and financial distress in a group firm.

Model (4) of Table 12 shows the regression results for CARs on the receipt of inter-corporate loans on loan variables, with no variables being statistically significant. The CARs are lower for loans with higher interest rate spreads, and these are significant at the ten percent level when we control for counter-party variables in Model (5) and further the firm characteristics in Model (6). A wider interest rate spread shows a higher default risk in the borrowing firm, which may attenuate the certification effect for these non-financial corporate lenders. However, there is no difference between intra-group and inter-group lenders for the market reactions to the receipt of inter-corporate loans.

State-owned lenders are associated with lower CARs in Model (6), which is significant at the five percent level and shows that these state-owned lenders are less credit-worthy in terms of certification for the borrowing firm. In contrast, state controlled firms are associated with higher CARs when receiving inter-corporate loans. State controlled firms are often worse in terms of performance, and inter-corporate loans may have a larger certification effect for such borrowing firms. Other accounts payable, however, is insignificant though the sign is positive in Model (6) of Table 12. Our results are generally consistent with the bank loan announcement literature in that certification is more effective if given by more credit-worthy lenders (Billett, Flannery and Garfinkel (1995)), and also more effective for poorly performing borrowers (Fields, Fraser, Berry and Byers (2006)).

In sum, we find that the CARs on issuance and receipt of inter-corporate loans are associated with various loan, counter-party, and (loan-announcing) firm variables. On the one hand, the issuance of inter-corporate loans generates lower CARs for loans with wider interest rate spreads, for intra-group loans, for firms with a higher market-to-book ratios, and with higher other accounts receivable (i.e., higher issuances of inter-corporate loans outstanding). On the other hand, the receipt of inter-corporate loans generates lower CARs for the loans with a higher interest rate spread, loans from state-owned lenders, and loans to non-state controlled borrowers. The results support the hypothesis that the issuance of inter-corporate loans signals to uninformed investors a lack of worthwhile projects to finance in the issuing firms, while the issuance of intra-group loans conveys additional information of financial distress in a group firm.

7. Post-performance in the long-run

7.1 Profitability

The changes in the performance ex post the inter-corporate loans can cast light on their long-term wealth effect. We examine firms' accounting performance after the inter-corporate loan announcements. If the issuance of inter-corporate loans reveals credit misallocation in the issuing firms or financial distress in a group firm, we would expect the accounting performance to turn worse ex post. In contrast, if the receipt of inter-corporate loans provides certification for the receiving firms or corporate propping-up, we would expect a higher accounting performance ex post. Table 13 shows the return on asset (ROA) in the years before, during, and after inter-corporate loan announcements. We find that ROA indeed declines significantly after the issuance of inter-corporate loans from one year before to one year afterwards, which confirms a lack of worthwhile projects to finance in the issuing firms. The results are qualitatively similar when we look at the change from one year afterwards.

[Table 13 here]

In addition, we find that ROA decreases for firms with relatively higher market to book ratio (i.e. above sample median) but not for firms with lower market to book ratio. A t-test between the two groups of firms shows that ROA decreases more from one year before to one/two years after inter-corporate loan announcements for firms with high market to book ratio, which are both significant at the ten percent level. Firms with higher growth potential indeed perform worse after the issuance of inter-corporate loans, which is consistent with a revelation for a lack of worthwhile projects to finance in the issuing firms. Besides, we find that ROA decreases significantly after the issuance of intra-group loans while it decreases less for inter-group loans, though the t-tests for the mean difference are not statistically significant. It seems that the issuance of intra-group loans reveals a lack of worthwhile projects to finance and additionally the default risks within the business group.

We find that ROA increases from the announcement year to two years after the receipt of inter-corporate loans, in particular for intra-group loans in Table 13. However, there is no significant difference between receiving firms with high versus low market to book ratio, or intra-group versus inter-group loans. In contrast, a t-test between the issuing versus receiving firms show that ROA decreases more for firms issuing inter-corporate loans from the announcement year to two years afterwards for loan announcing firms with both relatively high and low market to book ratio. However, ROA decrease more for intra-group loans from the announcement year to two years afterwards and significant at the one percent level, while insignificant for the inter-group loans.

7.2 Corporate investment

Changes of corporate investment behaviors may also shed some light on the wealth effect of inter-corporate loans. We explore the changes of capital expenditures scaled by the total assets in the year before the loan announcements for both the issuing and receiving firms. Table 14 shows the changes of capital expenditures in the years before and after the issuance of inter-corporate loans scaled by the total asset in the year before the loan announcements. We find that the capital expenditure increases less for the issuing firms than the receiving firms. In particular, this difference is significant for firms with relatively higher market to book ratios, and firms with

inter-group loans. However, we do not find any significant difference between issuers and receivers for firms with low market to book ratios, and firms with intra-group loans. Put it differently, firms issuing inter-corporate loans indeed invest less ex post, which confirms a lack of investment opportunities in the issuers and potential default risks in group firms.

[Table 14 here]

8. Conclusion

The granting of entrusted loans in China provides us with a unique setting to evaluate the valuation effects of inter-corporate lending and borrowing. We find that investors react negatively to the issuance of inter-corporate loans but positively to their receipt. The issuing of inter-corporate loans may indicate credit misallocation to uninformed investors, i.e., the issuing firms run out of worthy projects to finance. Furthermore, the issuance of inter-corporate loans to subsidiaries may reveal financial distress in that subsidiary, which may eventually spill over to the rest of the business group.

On the other hand, the receipt of intra-group loans, especially those from controlling shareholders, provides certification for the borrowing firms. In contrast to bank loan announcements, which often provoke negative market reactions (Bailey, Huang and Yang (2011)), non-financial corporate lenders in China can convey proprietary information to the uninformed investors.

We also confirm our results by linking the CARs to loan, counter-party, and firm level variables. The issuance of inter-corporate loans generate lower CARs for the intra-group loans, for loans with higher interest rate spreads, and for loans granted by young issuers, with higher market-to-book ratios and larger outstanding issuances of inter-corporate loans. In contrast, the receipt of inter-corporate loans generates lower CARs on loans with higher interest rate spreads, on loans from state-owned lenders, and on loans to non-state-controlled borrowers. Our results shed light on inter-corporate loans as signaling devices for credit misallocation by issuing firms, and for financial distress in a borrowing subsidiary, which can spill over to the rest of the business group.

We further examine the ex post performance of inter-corporate loans. Firms issuing inter-corporate loans have lower accounting performance, in particular for firms with higher growth potential and firms issuing intra-group loans. The issuing firms increase the corporate investment to a less extent than the receiving firms, which suggests a credit misallocation in the issuers of inter-corporate loans.

Although the inter-corporate loans have played an increasingly important role in China, we still know little about the welfare gains from these loans. Because inter-corporate loans rarely arise between listed firms, we cannot calculate the net gains from such loans. Further research on the net gains from inter-corporate loans would provide more insight on whether or not such loans should be encouraged.

References

- Allen, F., J. Qian, and M. Qian, 2005, "Law, Finance and Economic Growth in China," *Journal of Financial Economics* 77, 57-116.
- Allen, F., J. Qian, C. Zhang, and M. Zhao, 2013, "China's Financial System: Opportunities and Challenges," in J. Fan and R. Morck, eds, *Capitalizing China*, University of Chicago Press, Chicago IL.
- Allen, F., Y. Qian, G. Tu, and F. Yu, 2016, Entrusted Loans: A Close Look at China's Shadow Banking System, Imperial College, London, Mimeo.
- Almeida, H., C.-S. Kim, and H. B. Kim, 2015, "Internal Capital Markets in Business Groups: Evidence from the Asian Financial Crisis," *Journal of Finance* 70, 2539-2586.
- Almeida, H. V., and D. Wolfenzon, 2006, "A Theory of Pyramidal Ownership and Family Business Groups," *Journal of Finance* 61, 2637-2680.
- Andre, P., R. Mathieu, and P. Zhang, 2001, "A Note On: Capital Adequacy and the Information Content of Term Loans and Lines of Credit," *Journal of Banking and Finance* 25, 431-444.
- Arnott, R., and J. E. Stiglitz, 1991, "Moral Hazard and Nonmarket Institutions: Dysfunctional Crowding Out of Peer Monitoring?," *American Economic Review* 81, 179-190.
- Bae, K.-H., and G. K. Vidhan, 2009, "Creditor Rights, Enforcement and Bank Loans," *Journal of Finance* 64, 823-860.
- Bailey, W., W. Huang, and Z. Yang, 2011, "Bank Loans with Chinese Characteristics: Some Evidence on Inside Debt in a State-Controlled Banking System," *Journal of Financial and Quantitative Analysis* 46, 1795-1830.
- Beasley, M. S., 1996, "An Empirical Analysis of the Relation between the Board of Director Composition and Financial Statement Fraud," *Accounting Review* 71, 443-465.
- Best, R., and H. Zhang, 1993, "Alternative Information Sources and the Information Content of Bank Loans," *Journal of Finance* 48, 1507-1522.
- Billett, M. T., M. J. Flannery, and J. A. Garfinkel, 1995, "The Effect of Lender Identity on a Borrowing Firm's Equity Return," *Journal of Finance* 50, 699-718.
- Boissay, F., and R. Gropp, 2014, "Payment Defaults and Interfirm Liquidity Provision," *Review of Finance* Forthcoming.
- Boot, A. W. A., 2000, "Relationship Banking: What Do We Know?," *Journal of Financial Intermediation* 9, 3-25.
- Bose, P., 1998, "Formal–Informal Sector Interaction in Rural Credit Markets," *Journal of Development Economics* 56, 265-280.
- Buchuk, D., B. Larrain, F. Muñoz, and F. Urzúa I., 2014, "The Internal Capital Markets of Business Groups: Evidence from Intra-Group Loans," *Journal of Financial Economics* Forthcoming.
- Burkart, M., T. Ellingsen, and M. Giannetti, 2011, "What You Sell Is What You Lend? Explaining Trade Credit Contracts," *Review of Financial Studies* 24, 1261-1298.
- Calomiris, C. W., R. Fisman, and Y. Wang, 2010, "Profiting from Government Stakes in a Command Economy: Evidence from Chinese Asset Sales," *Journal of Financial Economics* 96, 399-412.

- Chen, K., J. Ren, and T. Zha, 2016, What We Learn from China's Rising Shadow Banking: Exploring the Nexus of Monetary Tightening and Banks' Role in Entrusted Lending, National Bureau of Economic Research, Cambridge MA, Working Paper.
- Claessens, S., S. D. Djankov, and L. H. P. Lang, 2000, "The Separation of Ownership and Control in East Asian Corporations," *Journal of Financial Economics* 58, 81-112.
- Claessens, S., J. Fan, and L. Lang, 2006, "The Benefits and Costs of Group Affiliation: Evidence from East Asia," *Emerging Markets Review* 7, 1-26.
- Degryse, H., M. Kim, and S. Ongena, 2009, *Microeconometrics of Banking: Methods, Applications and Results*, Oxford University Press.
- Ellingsen, T., T. Jacobson, and E. von Schedvin, 2016, Inter-Firm Lending: Contract-Level Evidence Contradicts Current Theories, Riksbanken, Stockholm, Working Paper.
- Fama, E. F., 1985, "What's Different about Banks?," *Journal of Monetary Economics* 15, 5-29.
- Fan, J. P. H., L. Jin, and G. Zheng, 2014, "Revisiting the Bright and Dark Sides of Capital Flows in Business Groups," *Journal of Business Ethics* 1-20.
- Fields, L. P., D. R. Fraser, T. L. Berry, and S. Byers, 2006, "Do Bank Loan Relationships Still Matter?," *Journal of Money, Credit, and Banking* 38, 1195-1209.
- Firth, M., C. Lin, and S. M. L. Wong, 2008, "Leverage and Investment under a State-owned Bank Lending Environment: Evidence from China," *Journal of Corporate Finance* 14, 642-653.
- Friedman, E., S. Johnson, and T. Mitton, 2003, "Propping and Tunneling," Journal of Comparative Economics 31, 732-750.
- Garcia-Appendini, E., and J. Montoriol-Garriga, 2014, "Firms as Liquidity Providers: Evidence from the 2007–2008 Financial Crisis," *Journal of Financial Economics* 109, 272-291.
- Gopalan, R., V. Nanda, and A. Seru, 2007, "Affiliated Firms and Financial Support: Evidence from Indian Business Groups," *Journal of Financial Economics* 86, 759-795.
- Harvey, C. R., K. V. Lins, and A. H. Roper, 2004, "The Effect of Capital Structure When Expected Agency Costs Are Extreme," *Journal of Financial Economics* 74, 3-30.
- He, Q., and O. M. Rui, 2014, "Ownership Structure and Insider Trading: Evidence from China," *Journal of Business Ethics* 1-22.
- Hoff, K., and J. E. Stiglitz, 1997, "Moneylenders and Bankers: Price-increasing Subsidies in a Monopolistically Competitive Market," *Journal of Development Economics* 52, 429-462.
- Huang, W., A. Schwienbacher, and S. Zhao, 2012, "When Bank Loans are Bad News: Evidence from Market Reactions to Loan Announcements under the Risk of Expropriation," *Journal of International Financial Markets, Institutions and Money* 22, 233-252.
- Jacobson, T., and E. Von Schedvin, 2015, "Trade Credit and the Propagation of Corporate Failure: An Empirical Analysis," *Econometrica* 83, 1315-1371.
- James, C., and D. C. Smith, 2000, "Are Banks Still Special? New Evidence on Their Role in the Corporate Capital Raising Process," *Bank of America - Journal of Applied Corporate Finance* 13, 52-63.

- James, C. M., 1987, "Some Evidence on the Uniqueness of Bank Loans," *Journal of Financial Economics* 19, 217-235.
- Jian, M., and T. J. Wong, 2010, "Propping through Related Party Transactions," *Review of Accounting Studies* 15, 70-105.
- Jiang, G., C. M. C. Lee, and H. Yue, 2010, "Tunneling through Intercorporate Loans: The China Experience," *Journal of Financial Economics* 98, 1-20.
- Johnson, S., R. La Porta, F. Lopez-de-Silanes, and A. Shleifer, 2000, "Tunneling," American Economic Review 90, 22-27.
- Khanna, T., and Y. Yafeh, 2005, "Business Groups and Risk Sharing around the World," *Journal of Business* 78, 301-340.
- La Porta, R., F. Lopez-de-Silanes, A. Shleifer, and R. W. Vishny, 1997, "Legal Determinants of External Finance," *Journal of Finance* 22, 1131-1150.
- Li, C., and S. Ongena, 2015, "Bank Loan Announcements and Borrower Stock Returns Before and During the Recent Financial Crisis," *Journal of Financial Stability* 21, 1-12.
- Love, I., L. A. Preve, and V. Sarria-Allende, 2007, "Trade Credit and Bank Credit: Evidence from Recent Financial Crises," *Journal of Financial Economics* 83, 453-469.
- Lummer, S. L., and J. J. McConnell, 1989, "Further Evidence on the Bank Lending Process and the Capital Market Response to Bank Loan Agreements," *Journal of Financial Economics* 25, 99-122.
- Maskara, P. K., and D. J. Mullineaux, 2011, "Information Asymmetry and Self-selection Bias in Bank Loan Announcement Studies," *Journal of Financial Economics* 101, 684-694.
- Megginson, W. L., A. B. Poulsen, and J. F. J. Sinkey, 1995, "Syndicated Loan Announcements and the Market Value of the Banking Firm," *Journal of Money, Credit and Banking* 27, 457-475.
- Mikkelson, W. H., and M. M. Partch, 1986, "Valuation Effects of Security Offerings and the Issuance Process," *Journal of Financial Economics* 15, 31-60.
- Ongena, S., and V. Roscovan, 2013, "Bank Loan Announcements and Borrower Stock Returns: Does Bank Origin Matter?," *International Review of Finance* 13, 137-159.
- Ongena, S., V. Roscovan, W.-L. Song, and B. Werker, 2014, "Banks and Bonds: The Impact of Bank Loan Announcements on Bond and Equity prices," *Journal of Financial Management, Markets and Institutions* 2, 3-28.
- Ongena, S., and D. C. Smith, 2000, "Bank Relationships: a Review," in P. Harker and S. A. Zenios, eds, *The Performance of Financial Institutions*, Cambridge University Press, London.
- Peng, W. Q., K. C. J. Wei, and Z. Yang, 2011, "Tunneling or Propping: Evidence from Connected Transactions in China," *Journal of Corporate Finance* 17, 306-325.
- Petersen, M. A., and R. G. Rajan, 1997, "Trade Credit: Theories and Evidence," *Review of Financial Studies* 10, 661-691.
- Poncet, S., W. Steingress, and H. Vandenbussche, 2010, "Financial Constraints in China: Firm-Level Evidence," *China Economic Review* 21, 411-422.
- Scharfstein, D. S., and J. C. Stein, 2000, "The Dark Side of Internal Capital Markets: Divisional Rent-Seeking and Inefficient Investment," *Journal of Finance* 55, 2537-2564.

- Stein, J. C., 1997, "Internal Capital Markets and the Competition for Corporate Resources," *Journal of Finance* 52, 111-133.
- Stiglitz, J. E., 1990, "Peer Monitoring and Credit Markets," *World Bank Economic Review* 4, 351-366.
- Thompson, R., 1985, "Conditioning the Return-Generating Process on Firm-Specific Events: A Discussion of Event Study Methods," *Journal of Financial and Quantitative Analysis* 20, 151-168.
- Wang, Y., and H. Xia, 2014, "Do Lenders Still Monitor When They Can Securitize Loans?," *Review of Financial Studies* 27, 2354-2391.
- Yook, K. C., 2003, "Larger Return to Cash Acquisitions: Signaling Effect or Leverage Effect?," *Journal of Business* 76, 477-449.

Variable category	Variable name	Definition		
Loan	Loan size	The amount of inter-corporate loan, in millions of RMB		
	Spread	The annual interest rate on the inter-corporate loan over the basis lending rate minus one, i.e., the interest premium over the basis lending rate		
	Maturity	The maturity of the inter-corporate loan, in months		
	Guarantee	=1 if the loan is collateralized or guaranteed by a third-party, 0 otherwise		
	Loan revision	=1 if the loan terms are revised, 0 otherwise		
Counter-party Firm	Intra-group loan	=1 if the counter-party firm is a loan-announcing firm's subsidiary, controlling shareholder, or belongs to the same business group		
	Counter-party ownership	A loan-announcing firm's equity ownership in the counter party firm, or the controlling shareholder's equity ownersh in a loan-announcing firm		
	Counter-party industry	=1 if the industry of the counter-party firm is the same as the loan-announcing firm, 0 otherwise		
	Counter-party size	The logarithm of the total assets of the counter-party firm		
	State-owned counter-party	=1 if the counter-party firm is state-owned, 0 otherwise		
(Loan-	Firm size	The logarithm of total assets		
announcing) Firm	Sales growth	The annual sales` growth rate		
	Market to book ratio	The ratio of the market value over the book value of assets		
	Cash holding	Cash over total assets		
	Free cash flow	Operating cash flow minus capital expenditure over total assets		
	Leverage	Total liabilities over total assets		
	State control	=1 if the ultimate owner is the state, 0 otherwise		
	Other accounts receivable	Other accounts receivable over total assets		
	Other accounts payable	Other accounts payable over total assets		

Table 1: Definitions of variables.

Panel A: Distribution of inter-corporate loan announcements by year and type						
Year	All	Issuance	Receipt			
2005	15	11	4			
2006	26	18	8			
2007	39	31	8			
2008	91	74	17			
2009	93	73	20			
2010	123	90	33			
2011	180	133	47			
2012	152	134	18			
Total	719	564	155			

 Table 2: Descriptive statistics on the announcement of inter-corporate

Panel B: I	Distribution of	f inter-corpora	te loan ann	ouncements	bv industrv

Industry names	All	Issuance	Receipt
Agriculture, forestry, animal husbandry and fishery	9	9	0
Mining	43	41	2
Manufacturing	337	264	73
Utilities	67	55	12
Construction	10	7	3
Transportation	29	22	7
Information technology	28	26	2
Wholesale and retail trade	51	48	3
Real estate	76	35	41
Social service	46	42	4
Communication and culture	5	5	0
Comprehensive	18	10	8
Total	719	564	155

Table 3: Characteristics of inter-corporate loans.

Loan size is the amount of inter-corporate loans in millions of RMB; Maturity is the loan maturity in numbers of months; Spread is the percentage increase in the interest rate from the basis lending rate; Guarantee equals one if a loan is collateralized or guaranteed by a third party, zero otherwise. Loan revision equals one if the loan terms are revised, zero otherwise. The test of mean difference between issuance and receipt reports the t-statistics with significance *** at one percent, ** at five percent, and * at ten percent level.

	Issuance Receip			Receipt		Test of mean difference	
	# Obs.	Mean	Median	# Obs.	Mean	Median	T-stat
			All	loans			
Loan size	564	185.23	70	155	300.26	180	-2.59***
Maturity	500	16.29	12	146	18.99	12	-2.14**
Spread	452	0.45	0.06	143	0.05	0	6.24***
Guarantee	564	0.29	0	155	0.12	0	4.19***
Loan revision	564	0.11	0	155	0.10	0	0.47
			Intra-gro	oup loans	•		
Loan size	396	216.57	80	140	298.28	150	-1.50
Maturity	339	18.22	12	134	19.07	12	-0.57
Spread	312	0.14	0	132	0.04	0	2.44**
Guarantee	396	0.13	0	140	0.09	0	1.05
Loan revision	396	0.11	0	140	0.11	0	0.05
			Inter-gro	oup loans			
Loan size	168	111.36	60	15	318.80	200	-3.97***
Maturity	161	12.24	12	12	18.17	12	-2.44**
Spread	140	1.15	1.21	11	0.17	0	3.79***
Guarantee	168	0.66	1	15	0.40	0	2.03**
Loan revision	168	0.11	0	15	0	0	1.38
Panel B: subsam	ple of int	ra-group	loans				
		To/fror	n control	ling shar	eholders		
Loan size	4	189.50	340	105	305.09	200	-0.55
Maturity	4	9.75	12	104	19.09	12	-1.19
Spread	4	0.12	0	102	0.01	0	0.79
Guarantee	4	0	0	105	0.10	0	-0.64
Loan revision	4	0	0	105	0.08	0	-0.57
		Т	o/from s	ubsidiari	es		
Loan size	375	216.93	80	10	165.50	80	0.27
Maturity	324	18.51	12	9	12.67	12	1.20
Spread	294	0.12	0	8	0.18	0	-0.40
Guarantee	375	0.12	0	10	0	0	1.16
Loan revision	375	0.11	0	10	0	0	1.09
	To bo	rrowers/fi	om lende	ers with c	other relation	ionship	
Loan size	16	225.25	60.10	25	322.80	100	-0.49
Maturity	10	12.60	12	21	21.71	12	-1.64*
Spread	13	0.40	0.10	22	0.11	0	1.26
Guarantee	16	0.31	0	25	0.12	0	1.52
Loan revision	16	0.19	0	25	0.28	0	-0.66

Panel A: all inter-corporate loans

Table 4: The determinants of issuance / receipt of inter-corporate loans.

The dependent variable is loan issuance / receipt which equals 1 if a firm issues/receives an inter-corporate loan, 0 otherwise. Firm age is the logarithm of the firm age; Market to book ratio is market value of equity plus book value of total liabilities scaled by book value of total assets; Leverage is total liabilities over total assets; State-control equals one if the ultimate controller of the firm is state-owned, zero otherwise. CEO also chairman equals 1 if the CEO is also the chairman of the board, 0 otherwise. Coefficients are reported with robust standard errors in parentheses. Significance indicated as *** at one, ** at five, and * at ten percent level.

	(1)	(2)	(3)	(4)	(5)	(6)
		Issuance			Receiving	
	Full sample	Intra-group	Inter-group	Full sample	Intra-group	Inter-group
Firm age	-0.01	-0.03**	0.03*	-0.01	0.01	-0.19**
	(0.01)	(0.01)	(0.02)	(0.02)	(0.02)	(0.08)
Market to book ratio	-0.07*	-0.04	-0.12**	0.03	-0.01	0.55***
	(0.04)	(0.05)	(0.05)	(0.04)	(0.05)	(0.18)
Leverage	-1.03***	-0.48	-2.24***	0.52	0.29	8.46*
	(0.16)	(0.32)	(0.38)	(0.67)	(0.69)	(4.98)
State control	0.37***	0.53***	0.13	0.34***	0.38***	-0.23
	(0.09)	(0.15)	(0.16)	(0.11)	(0.14)	(1.05)
EBIT	-0.59	-0.19	-1.44	0.77	1.19	-22.41*
	(1.04)	(1.18)	(1.84)	(2.07)	(2.27)	(11.64)
CEO also chairman	-0.40***	-0.82***	0.10	-0.30	-0.26	-1.25
	(0.12)	(0.32)	(0.20)	(0.21)	(0.18)	(0.81)
Constant	0.77***	0.56*	1.05***	-0.26	-0.38	-1.91
	(0.26)	(0.31)	(0.35)	(0.56)	(0.63)	(2.19)
Observations	486	317	181	161	136	25
Pseudo R2	0.049	0.075	0.109	0.025	0.025	0.357

Table 5: CARs on the issuance of inter-corporate loans.

Inter-corporate loans are classified as single-events if there is a single announcement for issuing a loan on the same date, and vice versa for multipleevents. The t-test of the CARs reports the t-statistic, and the Wilcoxon signedrank test report the proportion of positive CARs. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

	# Obs	Mean	Median	T-stat	Signed-rank test		
	All loans						
AR[0]	440	-0.37	-0.26	-4.14***	0.42***		
CAR[-1,0]	440	-0.42	-0.41	-3.08***	0.40***		
CAR[-1,1]	440	-0.49	-0.56	-2.74***	0.42***		
CAR[-2,2]	440	-0.70	-0.59	-3.02***	0.42***		
	Single event						
AR[0]	350	-0.40	-0.31	-4.00***	0.39***		
CAR[-1,0]	350	-0.48	-0.56	-3.07***	0.38***		
CAR[-1,1]	350	-0.51	-0.66	-2.52**	0.41***		
CAR[-2,2]	350	-0.67	-0.59	-2.53**	0.42***		
			Multiple	-events			
AR[0]	90	-0.26	0.12	-1.28	0.54		
CAR[-1,0]	90	-0.17	0.02	-0.65	0.50		
CAR[-1,1]	90	-0.41	-0.26	-1.08	0.44		
CAR[-2,2]	90	-0.81	-0.61	-1.72*	0.39*		

Table 6: CARs on the issuance for intra-group versus inter-group borrowers.

Inter-corporate loans are in the intra-group borrowers subsample if the borrowers are in the same business group, and vice versa for the inter-group borrowers subsample. The t-test of the CARs reports the t-statistic, and the Wilcoxon signed-rank test reports the proportion of positive CARs. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

	Intra-group borrowers							
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	222	-0.44	-0.29	-3.71***	0.41***			
CAR[-1,0]	222	-0.62	-0.43	-3.32***	0.39***			
CAR[-1,1]	222	-0.58	-0.54	-2.44**	0.43***			
CAR[-2,2]	222	-0.56	-0.47	-1.82*	0.44**			
			Inter-grou	p borrowers				
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	128	-0.32	-0.48	-1.81*	0.35***			
CAR[-1,0]	128	-0.25	-0.63	-0.88	0.36***			
CAR[-1,1]	128	-0.37	-0.73	-1.03	0.38**			
CAR[-2,2]	128	-0.86	-0.79	-1.75*	0.40**			

Table 7: CARs on the issuance of intra-group loans to controlling shareholders, subsidiaries, and borrowers with other relationships.

Intra-group loans are in subsamples according to controlling shareholders, subsidiaries, and borrowers with other relationships. The t-test of the CARs reports the t-statistic, and the Wilcoxon signed-rank test reports the proportion of positive CARs. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

	Intra-group loans to controlling shareholders							
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	4	-0.61	-0.55	-1.28	0.25			
CAR[-1,0]	4	-1.97	-0.83	-1.33	0.25			
CAR[-1,1]	4	-1.42	-1.32	-1.15	0.50			
CAR[-2,2]	4	-5.32	-4.13	-2.05	0.00*			
	Intra-group loans to subsidiaries							
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	206	-0.45	-0.25	-3.64***	0.42***			
CAR[-1,0]	206	-0.65	-0.43	-3.35***	0.38***			
CAR[-1,1]	206	-0.63	-0.54	-2.48**	0.43***			
CAR[-2,2]	206	-0.51	-0.47	-1.60	0.44**			
	Intra-	group loar	s to borrow	ers with other	relationships			
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	12	-0.17	-0.54	-0.35	-0.42			
CAR[-1,0]	12	0.43	-0.32	0.67	0.50			
CAR[-1,1]	12	0.44	-0.48	0.57	0.42			
CAR[-2,2]	12	0.22	0.45	0.20	0.50			

Table 8: CARs on the receipt of inter-corporate loans.

Inter-corporate loans are classed as single-events if there is a single announcement of a loan receipt on the same date, and as multiple-events if there are multiple announcements of loan receipts on the same date. The t-test of the CARs reports the t-statistic, and the Wilcoxon signed-rank test reports the proportion of positive CARs. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

···· p····.,	1 • • • •	,			
	# Obs	Mean	Median	T-stat	Signed-rank test
_			All lo	ans	
AR[0]	140	0.35	0.04	1.45	0.51
CAR[-1,0]	140	0.91	0.10	2.67***	0.51*
CAR[-1,1]	140	1.01	0.49	2.55**	0.54*
CAR[-2,2]	140	0.9	0.51	2.14**	0.56*
			Single	event	
AR[0]	125	0.46	0.18	1.74*	0.53
CAR[-1,0]	125	1.09	0.57	2.95***	0.54**
CAR[-1,1]	125	1.31	0.84	3.04***	0.58**
CAR[-2,2]	125	1.15	0.62	2.52**	0.60**
			Multiple	events	
AR[0]	15	-0.58	-0.52	-2.14*	0.33*
CAR[-1,0]	15	-0.59	-1.34	-0.86	0.33
CAR[-1,1]	15	-1.45	-2.02	-1.97*	0.20**
CAR[-2,2]	15	-1.15	-2.36	-1.27	0.27

Table	9: C	'ARs on	the recei	nt from	intra-groun	versus int	er-groun	lenders.
I abit	<i>.</i>		the recei	րլուտո	mu a-gi oup	versus me	ci-gioup	ichuci ș.

Inter-corporate loans are in the intra-group lenders subsample if the lenders are in the same business group, and as inter-group lenders if the lenders are not in the same business group. The t-test of the CARs reports the t-statistic, and the Wilcoxon signed-rank test reports the proportion of positive CARs. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

	Intra-group lenders							
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	110	0.43	0.18	1.69*	0.54			
CAR[-1,0]	110	1.06	0.62	2.79***	0.55**			
CAR[-1,1]	110	1.19	0.81	2.76***	0.59**			
CAR[-2,2]	110	1.04	0.6	2.12**	0.58*			
			Inter-gro	oup lenders				
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	15	0.68	-0.73	0.56	0.47			
CAR[-1,0]	15	1.33	-0.03	0.97	0.40			
CAR[-1,1]	15	2.16	1.42	1.26	0.53			
CAR[-2,2]	15	1.93	1.75	1.56	0.73			

Table 10: CARs on the receipt of intra-group loans from controlling shareholders, subsidiaries, and lenders with other relationship.

Intra-group loans are in subsamples according to controlling shareholders, subsidiaries, and borrowers with other relationship. The t-test of the CARs reports the t-statistic, and the Wilcoxon signed-rank test reports the proportion of positive CARs. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

	Intra-group loans from controlling shareholders							
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	81	0.55	0.30	1.99*	0.58			
CAR[-1,0]	81	1.42	0.72	3.07***	0.58***			
CAR[-1,1]	81	1.70	0.85	3.12***	0.60***			
CAR[-2,2]	81	1.54	1.15	2.53**	0.60**			
	Intra-group loans from subsidiaries							
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	7	-0.6	-1.14	-1.13	0.29			
CAR[-1,0]	7	-0.98	-1.35	-0.96	0.43			
CAR[-1,1]	7	-1.15	-1.75	-0.93	0.43			
CAR[-2,2]	7	-2.23	-1.96	-1.46	0.43			
	Int	ra-group le	oans from le	nders with other	relationship			
	# Obs	Mean	Median	T-stat	Signed-rank test			
AR[0]	22	0.28	-0.06	0.39	0.45			
CAR[-1,0]	22	0.37	-0.08	0.51	0.50			
CAR[-1,1]	22	0.04	0.72	0.07	0.59			
CAR[-2,2]	22	0.28	0.34	0.33	0.55			

Table 11: CARs sorted by loan, counter-party and firm characteristics.

Loan size is the amount of inter-corporate loans in millions of RMB; Maturity is the loan maturity in number of months; Spread is the percentage increase in the interest rate from the basis lending rate; Guarantee equals one if a loan is collateralized or guaranteed by a third party, zero otherwise; Loan revision equals one if a loan revises previous loan terms, zero otherwise; Counter-party ownership is the ownership of the controlling shareholder in a loan-announcing firm, or the ownership of a loan-announcing firm's subsidiary; Counter-party industry equals one if the counter-party and loan-announcing firm are in the same industry, zero otherwise; Counter-party size is the logarithm of total assets of the counter-party; State-owned counter-party equals one if the counter-party is state-owned, zero otherwise; Firm size is the logarithm of total assets; Age is the number of years listed on the stock exchanges; Sales growth is the annual sales growth rate; Market to book ratio is the market value of equity plus the book value of total liabilities scaled by the book value of total assets; ROA is the return on assets; Cash holding is cash over total assets; Free cash flow is the free cash flow over total assets; Counter sectivable is the other accounts receivable over total assets; Other accounts payable is the other accounts payable over total assets. The t-test of CAR difference between subsamples reports t-statistics with significance *** at one percent, ** at five percent, and * at ten percent level.

			Issua	suance of intragroup loans to subsidiaries		Issua	ance of inter-g	group loans	Receipt of intra-group loans from controlling shareholders		
Variable type	Variable name	Category	# Obs.	CAR[-1,0]	Difference of CAR	# Obs.	CAR[-1,0]	Difference of CAR	# Obs.	CAR[-1,0]	Difference of CAR
Loan	. .	> median	103	-0.0085***	0.0020	64	0.001	0.007	41	0.0161**	
	Loan size	\leq median	103	-0.0046*	-0.0039	64	-0.006	0.007	40	0.0123*	0.0038
	Samood	> median	69	-0.0044	0.0019	57	-0.0052	0.0070	18	0.0136	0.0005
	Spread	\leq median	95	-0.0062**	0.0018	56	0.0027	-0.0079	60	0.0141**	-0.0005
	Matarita	>1 year	51	-0.0055	0.0005	24	-0.0017	0.0003	23	0.0136	0.001
	Maturity	≤ 1 year	133	-0.0061**	0.0005	100	-0.0019	0.0002	57	0.0146***	-0.001
	Cuenentee	Yes	28	0.0055	0.0140**	84	-0.0055	0.0080	7	0.0059	0.0001
	Guarantee	No	178	-0.0084***	0.0140**	44	0.0033	-0.0089	74	0.0150***	-0.0091
	T	Yes	20	-0.0181**	0.0129*	18	-0.0003	0.0025	6	0.0238*	0.0102
	Loan revision	No	186	-0.0053***	-0.0128*	110	-0.0028	0.0025	75	0.0135***	0.0103
Counter-	Counter-party	> median	103	-0.0060**	0.0012				41	0.0201**	0.0120
party	ownership	\leq median	103	-0.0071**	0.0012				40	0.0082*	0.0120
	Counter-party	Same	109	-0.0067**	0.0001	17	-0.0061	0.0055	30	0.0129*	0.0014
	industry	Different	88	-0.0068**	0.0001	90	-0.0006	-0.0055	49	0.0114**	0.0014
	Counter-party	> median	89	-0.0063**	0.0012	40	0.0043	0.0081	27	0.0199**	0.0139
	size	\leq median	88	-0.0077**	0.0013	40	-0.0038		27	0.006	
	State-owned	Yes	150	-0.0064***	0.0012	30	0.0062	0.0100	58	0.0119**	-0.0083
	counter-party	No	54	-0.0077*	0.0013	94	-0.0047*	0.0109	23	0.0202**	
(Loan-	P :	> median	103	-0.0052*	0.0038	64	-0.0001	0.0040	41	0.0149***	0.0014
announcing)	Firm size	\leq median	102	-0.0080***	0.0028	64	-0.0049	0.0049	40	0.0135*	-0.0097
Firm		> median	103	-0.0043	0.0046	68	-0.0009	0.0024	45	0.0099**	
	Age	\leq median	102	-0.0089***	0.0046	60	-0.0043	0.0034	36	0.0196**	
	Color constitu	> median	101	-0.0050*	0.0027	61	-0.0001	0.0025	41	0.0209***	0.010
	Sales growth	\leq median	101	-0.0087***	0.0037	61	-0.0035	0.0035	38	0.0083	0.0126
	Market to book	> median	103	-0.0058*	0.0017	62	-0.0032	0.002	40	0.0199**	0.0000
	ratio	\leq median	102	-0.0074***	0.0016	62	-0.0012	-0.002	40	0.0100*	0.0099
	DOA	> median	103	-0.0063**	0.0007	64	-0.0017	0.0017	41	0.0224***	0.01/5*
	ROA	\leq median	102	-0.0069***	0.0006	64	-0.0033	0.0017	40	0.0059	0.0165*
	Cook holding	> median	103	-0.0055*	0.0021	64	-0.0003	0.0042	42	0.0155**	0.0025
	Cash holding	\leq median	102	-0.0076***	0.0021	64	-0.0047	0.0043	39	0.0129**	0.0025
	F 1.0	> median	99	-0.0098***	0.00(3	59	-0.0026	0.001	40	0.0042	0.01(0**
	Free cash flow	\leq median	98	-0.0035	-0.0063	59	-0.0016	-0.001	37	0.0210***	-0.0168**
	T	> median	103	-0.0057**	0.0017	65	-0.0001	0.0040	41	0.0091**	0.0104
	Leverage	\leq median	102	-0.0074**	0.0017	63	-0.005	0.0049	40	0.0195**	-0.0104
	G 1	Yes	155	-0.0060**	0.0024	80	-0.0015	0.0020	59	0.0124**	0.00/0
	State control	No	50	-0.0084**	0.0024	48	-0.0042	0.0028	22	0.0192*	-0.0068
	Other accounts	> median	103	-0.0098***	0.00/7*	64	-0.001	0.0020	41	0.0122**	0.0040
	receivable	\leq median	102	-0.0033	-0.0065*	64	-0.004	0.0030	40	0.0163**	-0.0040
	Other accounts	> median	103	-0.0090***	0.0040	68	0	0.0052	41	0.0170***	0.0057
	payable	\leq median	102	-0.0041	-0.0049	60	-0.0053	0.0053	40	0.0114	0.0056

Table 12: The regression of CAR[-1,0] on loan, counter-party and firm characteristics.

The dependent variable is the CAR[-1,0] in percentage points. Log loan size is the logarithm of the amount of inter-corporate loans in millions of RMB; Log maturity is the logarithm of the loan maturity in number of months; Spread is the percentage increase in the interest rate from the basis lending rate; Guarantee equals one if a loan is collateralized or guaranteed by a third party, zero otherwise; Loan revision equals one if a loan revises the previous loan terms, zero otherwise; Intra-group loan equals one if the counter-party is in a same business group, zero otherwise; Counter-party industry equals one if the counter-party and loan-announcing firm are in the same industry, zero otherwise; Counter-party size is the logarithm of total assets of the counter-party; State-owned counter-party equals one if the counter-party is state-owned, zero otherwise; Firm size is the logarithm of total assets; Age is the number of years listed in the stock exchanges; Market to book ratio is the market value of equity plus the book value of total liabilities scaled by the book value of total assets; Free cash flow is the free cash flow over total assets; Leverage is the total liabilities over total assets; State-control equals one if the ultimate controller of the firm is state-owned, zero otherwise; Other accounts receivable is the other accounts receivable over total assets; Other accounts payable is the other accounts payable over total assets. Industry and year dummies are included and the coefficients are omitted. Coefficients are reported with robust standard errors in parentheses. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

		Issuance	e of inter-c loans	orporate	Receipt of	inter-corp	orate loans
		(1)	(2)	(3)	(4)	(5)	(6)
Loan	Log loan size	-0.10	-0.04	-0.08	-0.52	0.01	1.16
		(0.17)	(0.23)	(0.26)	(0.54)	(0.65)	(1.15)
	Spread	-0.69**	-0.90*	-1.10**	-1.19	-5.44*	-11.42*
		(0.30)	(0.49)	(0.53)	(1.81)	(2.78)	(5.75)
	Log maturity	0.24	0.04	0.00	-0.50	-0.40	0.28
		(0.35)	(0.43)	(0.49)	(0.93)	(1.09)	(1.31)
	Guarantee	0.62	0.72	0.95	-0.31	0.93	2.91
		(0.43)	(0.54)	(0.62)	(1.47)	(2.30)	(2.91)
	Loan revision	-0.64	-1.04	-1.11	1.48	0.58	-1.10
		(0.76)	(0.90)	(0.92)	(1.55)	(1.48)	(2.35)
Counter-party	Intra-group loan		-1.64**	-1.74**		-0.20	-0.09
			(0.78)	(0.84)		(3.81)	(6.09)
	Counter-party industry		0.22	0.19		0.46	1.17
			(0.57)	(0.59)		(1.35)	(1.99)
	Counter-party size		-0.14	-0.23		0.20	-0.36
			(0.14)	(0.15)		(0.35)	(0.52)
	State-owned counter-party		0.73	0.72		-1.86	-7.40**
			(0.52)	(0.62)		(1.99)	(3.28)
(Loan-	Firm size			-0.08			-1.03
announcing)				(0.29)			(1.50)
Firm	Age			0.10*			-0.30
				(0.06)			(0.32)
	Market to book ratio			-0.42*			0.43
				(0.24)			(1.05)
	Free cash flow			-1.58			-9.27
				(1.37)			(7.30)
	Leverage			-1.16			-6.47
				(2.03)			(5.80)
	State control			-0.19			9.04*
				(0.59)			(4.70)
	Other accounts receivable			-14.49**			
				(7.16)			
	Other accounts payable						7.07
							(8.65)
	Constant	2.71	3.19	6.34	9.58	4.21	23.05
		(1.85)	(2.60)	(5.94)	(6.49)	(10.24)	(34.72)
	Industry and year dummie	Yes	Yes	Yes	Yes	Yes	Yes
	Observations	277	207	194	92	57	53
	R-squared	0.117	0.192	0.234	0.301	0.429	0.668

Table 13: ROA after inter-corporate loan announcements.

ROA is net income over total assets. [-1] / [0] / [+1] / [+2] indicates the number of years before / of / after the inter-corporate loan announcement. The t-test of the mean difference reports the t-statistic. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

		Issuance			Receipt	T-test of mean	
Summary statistics	N	Mean	St Dev	N	Mean	St Dev	difference
ROA[-1]	270	0 0459***	0.0040	03	0 0316***	0.0108	0.0143
	270	0.0459	0.0040	93	0.0310	0.0100	0.0145
ROA[+1]	270	0.0307	0.0033	92	0.0100	0.0117	0.0209
ROA[+1]	270	0.0350	0.0033	92	0.0240	0.0000	-0.0035
T-test of mean difference	270	0.0202	0.0041)2	0.0277	0.0055	0.0055
ROA[+1] - ROA[-1]	269	-0.0125***	0.0042	92	-0.0068	0.0121	-0.0057
ROA[+2] - ROA[-1]	269	-0.0192***	0.0048	92	-0.0017	0.0108	-0.0175*
ROA[+2]-ROA[0]	270	-0.0107***	0.0041	92	0.0204*	0.0119	-0.0310***
High market to book ratio	firm	8					
ROA[+1] – ROA[-1]	133	-0.0182***	0.0053	46	0.0012	0.0217	-0.0194
ROA[+2] – ROA[-1]	133	-0.0261***	0.0072	46	0.0002	0.0205	-0.0263
ROA[+2]–ROA[0]	133	-0.0134**	0.0061	46	0.0164	0.0181	-0.0298**
Low market to book ratio	firms						
ROA[+1] – ROA[-1]	132	-0.0031	0.0062	45	-0.0148	0.0112	0.0117
ROA[+2] – ROA[-1]	132	-0.0079	0.0061	45	-0.0032	0.0072	-0.0047
ROA[+2]-ROA[0]	132	-0.0071	0.0056	45	0.0252	0.0158	-0.0324**
Difference of high versus	low r	narket to bool	k ratio firm				
ROA[+1] – ROA[-1]		-0.0151*	0.0081		0.0160	0.0245	
ROA[+2] – ROA[-1]		-0.0182*	0.0094		0.0034	0.0219	
ROA[+2]-ROA[0]		-0.0063	0.0083		-0.0088	0.0241	
Intra-group loans							
ROA[+1] - ROA[-1]	171	-0.0153***	0.0050	78	-0.0027	0.0134	-0.0126
ROA[+2] - ROA[-1]	171	-0.0208***	0.0054	78	0.0002	0.0126	-0.0210*
ROA[+2]-ROA[0]	172	-0.0123**	0.0051	78	0.0242*	0.0139	-0.0364***
Inter-group loans							
ROA[+1] - ROA[-1]	98	-0.0077	0.0074	14	-0.0295	0.0276	0.0218
ROA[+2] - ROA[-1]	98	-0.0164*	0.0093	14	-0.0122	0.0101	-0.0041
ROA[+2]-ROA[0]	98	-0.0078	0.0069	14	-0.0006	0.0092	-0.0072
Difference of intra-group	versu	s inter-group	loans				
ROA[+1] - ROA[-1]		-0.0138	0.0087		0.0268	0.0337	
ROA[+2] - ROA[-1]		-0.0105	0.0101		0.0125	0.0302	
ROA[+2]-ROA[0]		-0.0046	0.0086		0.0248	0.0331	

Table 14: Capital expenditure after inter-corporate loan announcements.

CAPX is capital expenditure scaled by the total assets in the year before loan announcement. [-1] / [0] / [+1] / [+2] indicates the number of years before / of / after the inter-corporate loan announcement. The t-test of the mean difference reports the t-statistic. Significance is indicated as *** at one percent, ** at five percent, and * at ten percent level.

	Issuance				Receivin	T-test of mean difference	
Summary statistics	Ν	Mean	St.Dev	N	Mean	St.Dev	
CAPX[-1]	268	0.0550***	0.0035	92	0.0492***	0.0063	0.0058
CAPX[0]	269	0.0693***	0.0058	93	0.0739***	0.0209	-0.0046
CAPX[+1]	268	0.0786***	0.0068	87	0.1554**	0.0729	-0.0768*
CAPX[+2]	269	0.0812***	0.0070	91	0.1190***	0.0381	-0.0378
T-test of mean difference							
CAPX[+1] - CAPX[-1]	266	0.0239***	0.0061	86	0.104	0.0744	-0.0801*
CAPX[+2] - CAPX[-1]	267	0.0267***	0.0065	90	0.0719*	0.0388	-0.0451*
CAPX[+2] - CAPX[0]	269	0.0119**	0.0057	91	0.0447**	0.0220	-0.0328**
High market to book ratio f	irms						
CAPX[+1] - CAPX[-1]	130	0.0260***	0.0086	46	0.1957	0.1381	-0.1697**
CAPX[+2] - CAPX[-1]	131	0.0236***	0.0083	46	0.1161	0.0725	-0.0926**
CAPX[+2] - CAPX[0]	133	0.0035	0.0068	47	0.0536	0.0367	-0.0501**
Low market to book ratio fi	rms						
CAPX[+1] - CAPX[-1]	133	0.0241***	0.0087	40	-0.0014	0.0092	0.0255
CAPX[+2] - CAPX[-1]	133	0.0319***	0.0101	44	0.0256	0.0228	0.0063
CAPX[+2] - CAPX[0]	133	0.0208**	0.0093	44	0.0352	0.0234	-0.0145
T-test of high versus low ma	arket f	to book ratio	firms				
CAPX[+1] - CAPX[-1]		0.0019	0.0122		0.1971	0.1485	
CAPX[+2] - CAPX[-1]		-0.0083	0.0131		0.0906	0.0775	
CAPX[+2] - CAPX[0]		-0.0172	0.0115		0.0184	0.0442	
Intra-group loans							
CAPX[+1] - CAPX[-1]	172	0.0278***	0.0088	72	0.0425	0.0312	-0.0146
CAPX[+2] - CAPX[-1]	172	0.0334***	0.0092	76	0.0402*	0.0215	-0.0068
CAPX[+2] - CAPX[0]	174	0.0171**	0.0078	77	0.0318	0.0196	-0.0148
Inter-group loans							
CAPX[+1] - CAPX[-1]	94	0.0167***	0.006	14	0.4207	0.4308	-0.4040**
CAPX[+2] - CAPX[-1]	95	0.0146**	0.0071	14	0.2436	0.2215	-0.2290***
CAPX[+2] - CAPX[0]	95	0.0025	0.0077	14	0.1157	0.0945	-0.1132***
T-test of intra-group versus	inter-	group loans					
CAPX[+1] - CAPX[-1]		0.0019	0.1222		0.1971	0.1485	
CAPX[+2] – CAPX[-1]		-0.0083	0.0131		0.0906	0.0775	
CAPX[+2] – CAPX[0]		-0.0172	0.0115		0.0184	0.0442	

Appendix

Date	Туре	Name	Note	Institution
1992/12/30	Entrusted loan	A reply to the ICBC on the issues of entrusted loan	It clarifies several issues on entrusted loans correspondingly the request of the <i>Industrial and Commercial Bank of China</i> (ICBC), e.g. the definition of entrusted loans.	PBC
1996/5/16	Litigation guidance	A reply to Sichuan People's Higher Court on the qualification of subjects in the entrusted loan contracts	It specifies the rules for the subjects in the litigation cases on entrusted loan contracts corresponding a request by the <i>Sichuan People's Higher Court</i> .	PSC
1996/8/1	Entrusted loan	General rules on loans	It specifies detailed rules on entrusted loans.	PBC
1997/12/13	Litigation guidance	Issues on the litigation cases on certificates of deposit	It specifies several rules for the disputes in entrusted loan contracts.	PSC
1998/1/1	Disclosure requirements	Regulations of IPO by Shanghai and Shenzhen Stock Exchange	It specifies the disclosure requirement on entrusted loans, and also related party transactions. It has been revised seven times, and the latest version is effective from July 2012.	SSC and SZSE
1999/4/1	Interest rate lateralization	Interest rate regulations of RMB	It specifies the regulations for the interest rate of deposits and loans denominated in RMB.	PBC
2000/4/5	Entrusted loan	Notice on the issues for entrusted loan by commercial banks	It specifies a definition of entrusted loans, and switches the approval system to the registration system for entrusted loans.	PBC
2005/10/19	Entrusted loans	Notice on the CSRC suggestion on enhancing the quality of listed firms	It prohibits the entrusted loans from the listed firms to the controlling shareholders.	SCC
2005/10/27	Disclosure requirements	China securities law	It specifies the types of major events that should be announcement timely in Article of 67 at Chapter 3, e.g.	NPC
2007/2/2	Disclosure requirements	Explanatory notice on the regulations on information disclosure of listed firms	It specifies the information disclosure of extraordinary items for listed firms, e.g., entrusted loans.	CSRC
2013/7/19	Interest rate lateralization	Notice on the further reform for the marketization of interest rate	It lifts the regulation on the floor of the lending interest rate, and also the ceiling of the lending interest rate for rural credit cooperatives.	PBC
2013/12/10	Entrusted loan	Notice on several issues of tightening the regulation on shadow banking	It tightens the regulation for the shadow banking system including entrusted loans	SCC

Appendix 1: Timeline of laws and regulations related with entrusted loans.

Abbreviations: NPC is the National People's Council; PBC is People's Bank of China; PSC is the People's Supreme Court; SSC is the Shanghai Stock Exchange; SZSC is the Shenzhen Stock Exchange. SCC is the State Council of China; CSRC is the China Securities Regulatory Commission.

Appendix 2: Market shares of financing sources in China.

The upper panel shows the proportion of inter-corporate loan over total financing (left-axis, in percentage points) and the trend of total financing (right axis, in trillions of RMB), and the bottom panel shows the market share of various types of financing in the total financing of 2013. Data is retrieved from the PBOC website.





Appendix 3: Translation of an inter-corporate loan announcement.

Stock abbreviation: Guiyan Boye Stock code: 600459 No: Temporary 2011-4

Announcement of an entrusted loan to a fully owned subsidiary

The board of directors and all members declare that this announcement contains no false documentation, misleading statement or omission of important items, and bare individual and joint liability for the truthfulness, validity and completeness of the announcement.

Important notices for the entrusted loan

Financial institution: Kunming branch, China Citic Bank

Borrower: Guiyan Yimen Ziyuan Ltd (hereafter Yimen Ziyuan Ltd)

Amount: 30 million RMB

Maturity: One year

Interest rate: 7.07 percent per year

1. Summary

On Feb 25th, 2011, the eighth session of the fourth board meeting of the listed firm passes the proposal of providing an entrusted loan to a fully owned subsidiary. The board agrees to extend an entrusted loan of 30 million RMB to *Yimen Ziyuan Ltd*. This transaction does not constitute a related transaction. This entrusted loan does not need an approval from the shareholders' meeting.

2. Basic information about the borrower

Yimen Ziyuan Ltd is fully owned by the listed firm *Guiyan Boye*. It was set up on April 1st 2010 with the approval from the *Industrial and Commercial Administrative Bureau* of *Yimen County* at *Yunnan Province*. It has a registered capital of 50 million RMB, with the registered address: *Xiaolongkou Meishicheng, Xihuan Road, Longquan Town, Yimen County, Yuxi City, Yunnan Province*. Main businesses of the firm: the development and applications of the refinery skills for the resources of precious metals; the collection and processing of second-hand resources of precious metals; the manufacturing of basic products of precious metals; the manufacturing of special powder materials; the operation of skills and products made by the listed firm (according to the approved project and maturity if the operation involves special approvals by the laws).

Up until Sep 30th, 2010, the total assets of the *Yimen Ziyuan Ltd* is 54.76 million RMB; the total liabilities is 4.95 million RMB; total shareholders' equity is 49.81 million RMB; net profit is -188,000 RMB. None of the above numbers are audited by a third party.

3. Main content of the entrusted loan

According to the demand of *Yimen Ziyuan Ltd*'s operation and development, the listed firm provides an entrusted loan of 30 million RMB to *Yimen Ziyuan Ltd*. The loan has a maturity of one year, and an annual interest rate of 7.07 percent. (Please refer to the signed contract for the detailed items of the entrusted loan)

4. Sources of the fund for the entrusted loan

The fund is from the listed firm's self-owned fund. *Yimen Ziyuan Ltd* will repay the principal and interest in a lump sum at maturity.

5. Purpose of the entrusted loan and its effect on the listed firm

The entrusted loan will be used for *Yimen Ziyuan Ltd*'s operation and development. It will not affect the listed firm's normal operation as the fund is from the self-owned fund. *Yimen Ziyuan Ltd* is fully owned by the listed firm, so it can repay the entrusted loan at maturity.

Here announces the transaction.

The board of Guiyan Boye Co. Ltd., 2011/2/26

Appendix 4: The size of the entrusted loans in 2011 for firms listed on Shanghai Stock Exchange (in billions of RMB).

Data source: "An analysis on entrusted wealth management products and entrusted loans in 2011 for firms listed in Shanghai Stock Exchange" released by Shanghai Stock Exchange.

Categories	Balance 2010 year end	Loan issued	Loan received	Balance 2011 year end
Subsidiaries with a full or controlling ownership	52.77	65.56	30.25	88.08
Subsidiaries or joint ventures without a controlling ownership	4.18	5.15	2.87	6.47
Unconnected parties	7.26	14.52	8.34	13.44
Other connected parties	0.18	0.78	0.19	0.77
Total	64.38	86.01	41.64	108.75







Appendix 6: Summary statistics for loan, counter-party and firm characteristics.

Log loan size is the logarithm of the amount of inter-corporate loans in millions of RMB; Log maturity is the logarithm of the loan maturity in number of months; Spread is the percentage increase of the interest rate from the basis lending rate; Guarantee equals one if a loan is collateralized or guaranteed by a third party, zero otherwise; Loan revision equals one if a loan revises previous loan terms, zero otherwise; Intra-group loan equals one if the counter-party is in the same business group, zero otherwise; Counter-party industry equals one if the counter-party and listed firm are in the same industry, zero otherwise; Counter-party size is the logarithm of total assets of the counter-party; State-owned counter-party equals one if the counter-party is state-owned, zero otherwise; Firm size is the logarithm of total assets; Age is the number of years listed on the stock exchanges; Market to book ratio is the market value of equity plus the book value of total liabilities scaled by the book value of total assets; Free cash flow over total assets; Leverage is the total liabilities over total assets; State-control equals one if the ultimate controller of the firm is state-owned, zero otherwise; Other accounts receivable is the other accounts receivable over total assets; Other accounts payable is the other accounts receivable over total assets. The test of mean difference is conducted between firms, with issuance/receipt of inter-corporate loans versus firms without inter-corporate loan announcements. T-statistics are reported with significance *** at one percent, ** at five percent, and * at ten percent level.

		Issuance of inter- corporate loans		Receipt of inter- corporate loans		No inter-corporate loan firms			Test of mean difference (T-stat)				
		N	Mean	St.Dev	N	Mean	St.Dev	Ν	Mean	St.Dev	Issuance v.s. No loans	Receipt v.s. No loans	Issuance v.s. Receipt
Loan	Log loan size	338	4.39	1.26	103	4.95	1.31						-3.90***
	Spread	281	0.54	0.76	95	0.04	0.34						6.21***
	Log maturity	312	2.56	0.61	99	2.63	0.57						-1.02
	Guarantee	338	0.33	0.47	103	0.13	0.33						4.11***
	Loan revision	338	0.11	0.32	103	0.06	0.24						1.61
Counter-	Intra-group loan	338	0.62	0.49	103	0.85	0.35						-4.52***
party firm	Counter-party industry	308	0.42	0.49	96	0.41	0.49						0.27
	Counter-party size	261	1.81	1.77	68	4.73	2.10						-11.64***
	State-owned counter-party	332	0.55	0.50	103	0.67	0.47						-2.08**
(Loan-	Firm size	337	22.12	1.11	103	21.86	1.39	11,738	21.46	1.45	8.28***	2.78***	1.98**
announcing)	Age	337	10.95	4.71	103	11.34	5.09	10,926	9.61	5.16	4.71***	3.39***	-0.71
Firm	Market to book ratio	333	2.14	1.22	102	2.29	1.59	11,461	2.32	1.58	-2.01**	-0.18	-1.02
	Free cash flow	319	0.02	0.17	97	0.03	0.14	10,441	0.04	0.15	-2.29**	-0.74	-0.94
	Leverage	337	0.47	0.19	103	0.61	0.22	11,738	0.50	0.24	-2.69***	4.81***	-6.64***
	State control	337	0.71	0.45	103	0.67	0.47	11,050	0.51	0.50	7.24***	3.24***	0.76
	Other accounts receivable	337	0.03	0.04	103	0.02	0.03	11,580	0.04	0.06	-2.80***	-2.16**	0.82
	Other accounts payable	337	0.04	0.04	103	0.08	0.10	11,585	0.05	0.07	-4.07***	3.84***	-6.21***



CFS Working Paper Series

Recent Issues

All CFS Working Papers are available at www.ifk-cfs.de.

No.	Authors	Title
528	Christina E. Bannier and Milena Neubert	Actual and perceived financial sophistication and wealth accumulation: The role of education and gender
527	Dae Woong Kang, Nick Ligthart, and Ashoka Mody	The European Central Bank: Building a Shelter in a Storm
526	Peter A. Zadrozny	Extended Yule-Walker - Identification Of Varma Models With Single- Or Mixed Frequency Data
525	Christiane Baumeister, Lutz Kilian	Forty Years of Oil Price Fluctuations: Why the Price of Oil May Still Surprise Us
524	Roman Kräussl, Thorsten Lehnert, Sigita Senulyte	Euro Crash Risk
523	William Schulze, David Deeds, Robert Wuebker, Roman Kräussl	Toward a Pecking Order Theory of Strategic Resource Deployment
522	Ronald Bosman, Roman Kräussl, Elizaveta Mirgorodskaya	The "Tone Effect" of News on Investor Beliefs: An Experimental Approach
521	Otmar Issing	Completing the Unfinished House: Towards a Genuine Economic and Monetary Union?

Center for Financial Studies | Goethe University | Theodor-W.-Adorno-Platz 3 | 60323 Frankfurt am Main | Germany Phone +49 69 798 30050 | Fax +49 69 798 30077 | info@ifk-cfs.de | www.ifk-cfs.de